

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bortolotti, Bernardo; Milella, Valentina

Working Paper Privatization in Western Europe Stylized Facts, Outcomes, and Open Issues

Nota di Lavoro, No. 124.2006

Provided in Cooperation with: Fondazione Eni Enrico Mattei (FEEM)

Suggested Citation: Bortolotti, Bernardo; Milella, Valentina (2006) : Privatization in Western Europe Stylized Facts, Outcomes, and Open Issues, Nota di Lavoro, No. 124.2006, Fondazione Eni Enrico Mattei (FEEM), Milano

This Version is available at: https://hdl.handle.net/10419/74088

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Privatization in Western Europe Stylized Facts, Outcomes, and Open Issues Bernardo Bortolotti and Valentina Milella

NOTA DI LAVORO 124.2006

OCTOBER 2006

PRCG – Privatisation, Regulation, Corporate Governance

Bernardo Bortolotti, University of Turin and Fondazione Eni Enrico Mattei Valentina Milella, Fondazione Eni Enrico Mattei

This paper can be downloaded without charge at:

The Fondazione Eni Enrico Mattei Note di Lavoro Series Index: http://www.feem.it/Feem/Pub/Publications/WPapers/default.htm

Social Science Research Network Electronic Paper Collection: http://ssrn.com/abstract=936911

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei Corso Magenta, 63, 20123 Milano (I), web site: www.feem.it, e-mail: working.papers@feem.it

Privatization in Western Europe Stylized Facts, Outcomes, and Open Issues

Summary

Privatization has certainly been one of the main events of the economic and financial history of the 20th century. Between 1997 and 2004 more than 4,000 privatization operations were carried out in the world, bringing to governments revenues for over 1,350US\$billion. Western Europe emerges as the most important region, having implemented the greatest number of privatizations and raised a half of global revenues. The relevance of Western Europe in the process can be ascribed to several factors. This paper investigates the causes of this process, summarizes the main trends of privatization activity at the country level, analyzes the main privatization at the macro and microeconomic level.

Keywords: Privatization, State-Owned Sector, Capital Markets, Financial Development

JEL Classification: L33, L38

Report prepared for the Initiative for Policy Dialogue (IPD), Privatization Task Force, Columbia University. Research assistance from Luca Farinola is gratefully acknowledged.

Address for correspondence:

Bernardo Bortolotti Fondazione Eni Enrico Mattei C.so Magenta, 63 20123 Milano Italy E-mail: bernardo.bortolotti@feem.it

1. Introduction

Started in the United Kingdom at the end of the 1970s, privatization spreads in Continental Europe during the 1980s. Western European countries, hard pressed to improve mounting fiscal deficit and to introduce major product market reforms as requisite to join the European Union, pushed ahead during the 1990s one of the most extensive and ambitious privatization programs around the world.

After more than a quarter of century, it is possible to take stock of the main national experiences, and try to draw an overarching description of what has been achieved and not achieved through such a sustained divestiture policy.

The report presents quantitative information about the size and the extent of the State sell-offs and State ownership in major European countries, and analyzes the most important stylized facts about the causes and consequences of the process.

The main findings can be summarized as follows. Privatization in western Europe has been mainly driven by fiscal conditions and by the positive outlook in financial markets. The process has also been shaped by political preferences and institutional constraints, as partisan politics and constitutional rules affected privatization choices.

As to the consequences, our knowledge about the real effects of privatization in terms of fiscal consolidation and operating efficiency of firms is more limited, even if the impact of privatization on financial market development and on the spreading of equity culture has often been dramatic.

Surprisingly, the large scale privatization process of the 1990s did not alter the prevailing corporate governance structures in privatized firms. At the turn of the century, we find European governments firmly controlling (by voting rights and golden shares) a large part of privatized companies, especially in strategic sectors. Understanding whether the coexistence of private ownership and public control is a European transient anomaly or a functional pattern of governance is important for policy reasons and might be an exciting avenue for future research.

The report is organized as follows. Section 2 reports the main trends of privatization activity; section 3 describes the emergence of the State-owned enterprise (SOE) sector; section 4 analyzes the main privatization drivers; sections 5 provides an account of the main findings on the effects of privatization at the macro and microeconomic level. Section 6 concludes.

2. Privatization trends

The first aggregation of data – referred to the global number of transactions and revenues raised in the 1977-2004 period - provides a preliminary indication of the extent of privatization in Western Europe as compared to the rest of the world.

Western Europe appears to be the area mostly involved in the process, having implemented the greatest number of privatizations (29 percent of global deals) and raised 48 percent of global revenues (see Table 1, and Figure 1).

[INSERT ABOUT HERE Table 1, and Figure 1]

In terms of percentage of global proceeds, Western Europe is followed by Asia (24 percent) and Latin America (11 percent). As to the number of transactions, privatizations sales were also numerous in Central and Eastern Europe and the former Soviet Union (28 percent) but limited in size (representing only 6 percent of global revenues). The opposite occurred in Asia (see Table 1).

The data on the privatization methods adopted in Western Europe confirm a general trend, with private sales (henceforth PS), i.e. a private equity placement to strategic investors, accounting for the majority of cases. Privatization on public equity markets (public offerings, henceforth PO) are less frequent (28 percent) and typically raise higher revenues (64 percent), being used for larger and often more profitable companies which can be easily floated in domestic and/or international exchanges.

The vast extent of European privatization (see Figures 2, and 3) can be ascribed to the large size of the State-owned Enterprise sector (henceforth SOE) of most European economies, and to the exceptional weight of the British experience.

[INSERT ABOUT HERE Figures 2, 3]

Indeed, privatization was one of the building blocks of the Thatcherite reforms, which shrank the size of the SOE sector in the UK from 10 percent of GDP to virtually nil. The 1977 public offer of British Petroleum (BP) is usually considered - after the failed German attempts of the 1950s under the Adenauer government - the first large-scale privatization in modern times.

Shortly after, in the mid-1980s privatization started to spread out also in Continental Europe. In 1985 Italy undertook the long lasting process of denationalization of the State holding company IRI, with the partial sale of SIRTI and Alitalia, and in 1986 the newly elected French conservative government pushed ahead a highly politicized (re)privatization of its financial institutions.

In 1989 Portugal, Spain, the Netherlands and Sweden entered the process. Italy, Portugal and Turkey reported their first large-scale sales in 1993. Throughout the 1990s also Belgium, Greece and Ireland joined the process.

Privatizations experienced an exponential growth at the end of the 1990s, reporting a peak in revenues in 1998 due to a number of large PO in "strategic" sectors, such as ENI (petroleum) in Italy, Swisscom (tlc) in Switzerland, ENDESA (electric utility) in Spain, and France Telecom (tlc) in France. 1999 boasts a remarkable level of revenues, also thanks to the first tranche of the Italian electric generation company ENEL in October, which still represents today the largest IPO (initial public offer) in history.

At the turn of the century the process abruptly slowed down. Between 2000 and 2002 sales and revenues decreased at an average rate of 34 and 50 percent, respectively. Privatization activity in 2002 fell down to the levels reported at the initial stage of the cycle. After this striking dip, mainly due to the global economic downturn and negative stock market conditions, the process resumed in 2003 and regained momentum in 2004 (see Figure 4). Several large PO in the telecommunications (France Telecom, Deutsche Telekom, Telekom Austria, and Telia Sonera); in the financial sector (Eulia in France and Deutsche Postbank in Germany); in the oil and gas industry (French Total, and Norwegian Statoil) boosted privatization revenues to US\$59bn, a figure close to the historical peak levels of the European process.

Importantly, 2004 marked the resurgence of global offers and retail investors appetite towards privatized stocks, allowing for example the implementation of a large offering of shares of ENEL, the Italian electricity giant, which brought into the government coffer over US\$9.5bn.

[INSERT ABOUT HERE Figure 4]

A preliminary analysis of privatization trends at the national level shows that within European countries, the United Kingdom leads the ranking by total revenues (see Figure 5), and Italy boasts the second position, followed by Germany, France, and Spain.

However, revenues scaled by GDP provide a more appropriate measure for a cross country comparison. Indeed, since the size of a country matters in explaining the extent of privatization (as it also affects the size of the SOE sector), the final ranking changes considerably when total revenues are scaled by GDP (see Figure 5, and Figure 6), even if countries such as the UK and Italy remain in prominent positions.

[INSERT ABOUT HERE Figure 5, and 6]

[INSERT ABOUT HERE Table 2]

The comparison between the mean and median values of privatization revenues suggests that the distribution of total revenues is strongly affected by the presence of a few "deep" privatizing countries such as the UK, Italy, and Germany, which report values well above the sample mean. Italy boasts the highest value in average revenues, meaning that larger companies have been sold.

The ratio of public offers to total deals provides some information about the privatization methods. Austria and Portugal have opted more systematically for flotation in public equity markets, while in Germany, Belgium and Sweden the resort to share issues has been much more limited.

The breakdown by industry (see Figure 7) shows that almost no sector is left out of the privatization process. However, the greater part of revenues comes from telecommunications, utilities, the manufacturing industry, finance, petroleum and transports.²

² Industrial sectors are defined as follows. *Agriculture Industry* (SICs 01XX - 09XX) includes: Agricultural Production Crops; Agricultural Production Livestock; Agricultural Services; Forestry; Fishing, Hunting, and Trapping. *Natural Resource* (SICs 10XX;12XX; 14XX) includes: Metal

[INSERT ABOUT HERE Figure 7]

As a general rule, the initial stage of the process involves the manufacturing and industrial sectors, and the financial institutions (the latter in prominent position in terms of proceeds), while the privatizations of telecommunications, energy, transports and utilities (henceforth "strategic sectors") typically step into the second stage.

However, the timing of the entry of sectors has been different in Continental Europe with respect to the United Kingdom, and definitely the UK went furthest in the shortest time, privatizing in 1977 its national oil company British Petroleum (BP), the telecommunications in 1981 (with the first tranche of Cable & Wireless), several water and electric utilities all along the 1980s, and shortly after the railways.

In the other European countries, apart from some scattered cases at the end of the 1980s (i.e. Alitalia in 1985, OMV, Cie Général Téléphoniques and VEBA in 1987, and ENDESA in 1988), the telecommunications, the utilities, transports and energy sectors are still firmly in public hands until the first half of the 1990s.

Mining; Coal Mining; Nonmetallic Minerals, except Fuels. Petroleum Industry (SICs 13XX; 29XX) includes: Oil and Gas Extraction; Petroleum and Coal Products. Construction (SICs 15XX - 17XX) includes: General Building Contractors; Heavy Construction, Except Building; Special Trade Contractors. Finance (SICs 60XX - 67XX) includes: Depository Institutions; Non Depository Institutions; Security and Commodity Brokers; Insurance Carriers; Insurance Agents, Brokers, & Service; Real Estate; Holding and Other Investment Offices. Manufacturing (SICs 20XX - 28XX; 30XX - 39XX) includes: Food and Kindred Products; Tobacco Products; Textile Mill Products; Apparel and Other Textile Products; Lumber and Wood Products; Furniture and Fixtures; Paper and Allied Products; Printing and Publishing; Chemical and Allied Products; Rubber and Miscellaneous Plastic Products; Leather and Leather Products; Stone, Clay, and Glass Products; Primary Metal Industries; Fabricated Metal Products; Industrial Machinery and Equipment; Electronic and Other Electric Equipment; Transportation Equipment; Instruments and Related Products; Miscellaneous Manufacturing Industries. Public Administration (SICs 91XX - 97XX) includes: Executive, Legislative, & General; Justice, Public Order, & Safety; Finance, Taxation, & Monetary Policy; Administration of Human Resources; Environmental Quality & Housing; Administration of Economic Programs; National Security & International Affairs; Nonclassifiable Establishments. Services (SICs70XX- 89XX) includes: Hotels and Other Lodging Places; Personal Services; Business Services; Auto Repair, Services, and Parking; Miscellaneous Repair Services; Motion Pictures; Amusement & Recreation Services; Health Services; Legal Services; Educational Services; Social Services; Museums, Botanicals, Zoological Gardens; Membership Organizations; Engineering & Management Services; Private Households; Services, Misc. Telecommunications (SICs 48XX) includes: Communications. Trade Industry (SICs 50XX - 59XX) includes: Wholesale Trade - Durable Goods; Wholesale Trade - Nondurable Goods; Building Materials & Garden Supplies; General Merchandise Stores; Food Stores; Automotive Dealers & Services Stations; Apparel and Accessory Stores; Furniture and Home Furnishings Stores; Eating & Drinking Places; Miscellaneous Retail. Transportation Industry (SICs 40XX - 42XX; 44XX; 45XX; 47XX) includes: Railroads Transportation; Local and Interurban Passenger Transit; Trucking and Warehousing; Water Transportation; Transportation by Air; Transportation Services. Utilities (SICs 46XX; 49XX) includes: Pipelines, Except Natural Gas; Electric, Gas, and Sanitary Services.

Only from 1994 onwards, indeed, "strategic" sectors actually became involved in the process. Among the oil companies outside the United kingdom, Elf Aquitaine, ENI, OMV, Total, and Repsol are the first to be privatized during the early 1990s. Shortly after, coupled with the global context of technological innovation and trade liberalization pushed by WTO, several privatizations involved the telecommunications of Spain, Switzerland, Denmark, Finland, and the Netherlands.

During the second half of the decade, also water, electric utilities and transports started to be privatized, and the percentage of revenues raised trough the sales of public assets in "strategic" sectors appears to be highly increasing over time – especially between 1994 and 1997 – remaining then stable at quite high levels (well above 60 percent of yearly total revenues) from 1998 to date (see Figure 8).

[INSERT ABOUT HERE Figure 8]

Overall, the privatizations of strategic sectors raised 68 percent of European privatization revenues trough 38 percent of the total area transactions. However, despite these significant results, it seems that European countries fell short to accomplish ambitious privatization programs in strategic sectors, and only few countries, such as the UK and Spain have fully privatized strategic sectors such as energy, telecommunications, or transport, while in some other countries, despite some recent announcements, the majority of assets in strategic sectors (particularly in energy) are still publicly owned.

3. The State-Owned Enterprise Sector in Western Europe

The aggregate data presented provide a preliminary description of European divestitures. However privatization processes would be more fully understood by relating the extent of state sell-offs to the size of the State owned enterprises (SOE) sector pre-privatization.

To our knowledge, the only centralized source of comprehensive data on SOE activity is the World Bank (1995), reporting several indicators for industrialized and developing economies for the 1978–1991 period.³ SOEs are defined as "government

³ See also Nicoletti and Scarpetta (2003); Haggarty and Shirley (1995).

owned or controlled entities that generate the bulk of their revenues from selling goods and services."

We therefore used the WB database in order to build a proxy for initial conditions given by the value added of State-owned enterprises as a percentage of GDP in the year before the first operation reported by Securities Data Corporation databases (where possible), certainly the most comprehensive source for privatization information at the transaction level.

As shown in Table 2, on average, SOEs accounted for a significant fraction of the economic activity in Western Europe pre-privatization. The European average, indeed, is about 10 percent, while for non European OECD economies this figure narrows to approximately 7 percent. Several countries, such as the UK, Germany, France, Austria, Portugal and Sweden range between 10.

However, these data suffer from several drawbacks. First, the definition used by the World Bank limits to "commercial activities controlled by virtue of government's (direct and indirect) ownership stake alone." However, in several cases, SOEs are owned by regional and local bodies, which do not enter in the definition. In other cases, the database does not report the stakes owned by the government in financial entities, and does not take into account the government's indirect ownership.

For example, the Swedish SOE share of value added would have certainly been higher - and probably larger than the EU average – if the ownership of local and regional bodies were included. Local municipal governments and the regions/county councils control about 1,777 local enterprises altogether, operating especially in housing and energy sectors. The case of Portugal illustrates how the exclusion of the financial sector may distort the real size of the SOE sector. In the mid-1970s the State nationalized the nine largest banks and eight insurance companies. As a consequence, the State owned indirectly hundreds of small and medium enterprises, in which formerly private banks held controlling stakes (Baklanoff, 1986). Finally, the value added figures for Belgium only refer to transport and telecommunication sectors, largely underestimating the size of the SOE sector in a historically highly interventionist State.

Due to these limitations, World Bank data partially fail to appreciate the real size of State ownership in Europe, suggesting the need for a more systematic data collection to fill the gap. Albeit biased, the data show quite clearly that at the beginning of the 1980s State ownership of productive assets was very large. What factors explain the emergence of such a large SOE sector in Europe?

At the risk of oversimplification, the rise of the State ownership in Western Europe can be traced back to the 20th century, and particularly to three waves of nationalizations that occurred (i) after the Great Depression of 1929-1933, (ii) during the post-WWII period, and (iii) after the oil shocks in the mid 1970s.

The economic downturn caused by the Great Depression led to a strong interventionist approach almost everywhere. In the 1920s, the French and the Belgian governments established financial institutions taking control of the banking sector. In Germany, from the Weimar Republic to the National Socialist period, large scale nationalizations were implemented to foster the industrialization process. Similarly, important nationalizations took place in Austria, involving the telecommunication, transport and banking sectors.

Similarly, in 1933 - under the fascist era- the state-owned industrial holding Istituto per la Ricostruzione Industriale (IRI) was created in Italy in order to recover the national economy. In Spain, the root of State-owned industry dates back to the establishment of Franco's dictatorship. After the Civil War, Spain imported the "IRI" model, creating the Institudo Nacional de Industria (INI), with the aim to strengthen domestic development, foster import substitution, and inject growth in underdeveloped areas. In Portugal, since 1933, the "corporative" ideology became the manifesto of the Salazar's authoritarian regime, which aimed at keeping political and economic activity under tight public control.

As to Northern European countries, State-owned enterprises in Sweden have been established at the beginning of the 20th century to better exploit national resources and to in particular the coal, timber and steel industries. In Finland, the State's economic activities of the 1920s and 1930s have been spurred by the lack of private venture capital and mainly aimed at the exploitation of raw materials, development of infrastructure, achievement of self-sufficiency, financing of business activities and implementation of regional policies.

The second wave of nationalizations were implemented after World War II, and carried on within the economic reconstruction. In the UK, between 1945 and 1951, the SOE sector became one of the largest in Western Europe. In France, in the 1945-1946 period, the State took over control of economic sectors requiring heavy capital

injection - such as coal, electricity, gas and railways - along with credit and insurance sectors. At that time France established a centralized planning body, the Commissariat Général au Plan, with the mission to plan and co-ordinate the entire activity of the public sector. Belgium implemented important post-war nationalizations through the State holding company Societé Nationale d'Investissement (SNI), although the government already controlled indirectly a large number of enterprises through banks and credit institutions (Societé Générale de Belgique, Caisse Générale d'Epargne et de Retraite (CGER), Crédit Communal de Belgique). In Austria, the nationalization basically involved all the former already nationalized companies successively taken over by the German Reich. The former "German property", indeed, - including industrial enterprises founded by the Nazis accounted for one fifth of Austrian value added. It included the country's three largest banks, the entire coal and metal mining industry, all mineral oil extraction and processing facilities, and all the important companies in the heavy industry sector. Nationalized industry made the Austrian public sector one of the largest in continental Europe. The State holding company Öesterreichishce Industrieholding (ÖIAG) played an important role in economic reconstruction in the occupation zones, also thanks to the American reconstruction aid.

The third wave of nationalizations occurred in Europe in the aftermath of the oil shocks, in the period that goes from the mid 1970s to mid 1980s. State-owned firms were increasingly used for stabilization and direct employment policies, to rescue ailing private firms, sustain investment, and support underdeveloped regions. In general, and in particular in Spain, Italy and Sweden, non-economic goals were put above corporate policy objectives.

In France, the promotion of "national champions", which competed from a favored position with domestic and foreign rivals, has been an important feature of the industrial policy of the period. Along with the direct ownership of the main economic sectors, including key industries, the State control over the economy took the form of industrial protection and subsidies, and control over foreign investments. This whole range of policy instruments typified the "French Model" of State intervention in the economy. Finally, during the revolutionary period (1974-1976), Portugal launched a massive nationalization program, involving a large number of companies which came under direct State control via the holding company IPE (Investimentos e Participaçoes do Estado).

9

The abnormal growth of the SOE sector ensuing from these three waves of nationalization clashed with the rising requirements of global competitiveness. The pursued type of stabilization policy allowed to smooth the adjustment process, but caused negative effects on the productivity and profitability of SOEs and on the state of public finances.

In early 1980s, the problem of the inefficiency of the SOE sector – absorbing an increasing amount of public subsidies – became a priority in the political agenda of most European countries, prompting the surge of privatizations that began in the 1980s and gathered momentum from 1991 onwards after the ratification of the Maastricht Treaty. The restructuring and privatization of the SOE sector became necessary not only to modernize the economy, but also to meet convergence criteria without politically costly tax increases.

4. Determinants of privatization

The previous section has pointed out that common factors can explain the rise of the SOE sector in Europe, and that with a few exceptions, European governments, at the beginning of the 1980s, owned on average large chunks of the national economy. Within this common pattern, historical specifics matter and can be reflected in the cross country variability in the percentages of SOE value added that we observe in the European context.

Although some caution is needed in the interpretation of World Bank data, it is certainly interesting to construct an indicator about the extent of privatization in a given country which could take into account initial conditions. In this direction we have scaled the total revenues raised in a given country in the 1977-2000 period by the total valued added of SOE in the year preceding the first privatization sale (Rev/SOE in Table 2). The number obtained provides therefore a measure of the "size" of one country's privatization process relative to what governments have to sell.

The data show that Portugal has carried out the largest contraction in State property, followed by the UK, Netherlands. Spain and Italy rank in middle high position, while France and Austria report the lowest scores. Interestingly, as shown in Figure 9, privatization revenues scaled by our proxy for the size of the SOE sector appear to be strongly correlated with the revenues to GDP ratio (corr. = 0.87), a widely used measure in cross country analyses of the size of privatization. Rev/GDP series can be easily constructed for a large number of countries and therefore allow for panel data empirical analyses. The positive correlation with a more proper measure of one country's privatization effort is reassuring, and suggests the feasibility of comprehensive empirical analyses on the political and economic determinants of privatization in Europe.

[INSERT ABOUT HERE Figure 9]

Bortolotti and Pinotti (2003) perform an econometric analysis on a sample of 21 developed countries (16 European) for the 1977-1999 period, estimating in Tobit panel regressions the yearly ratio of revenues to GDP. Macroeconomic conditions appear particularly relevant. The extent of privatization appears higher in countries with higher per capita GDP and lower growth rates, which in turns means that privatization characterizes a relatively advanced stage of economic development. The inverse relation found between (lagged) growth rates and privatization could also indicate that governments tend to resort to privatization when the economic outlook deteriorates, in order to foster economic activity via an increase in private investment. Fiscal conditions are also particularly relevant, as we find the debt ratio always highly statistically significant. Indeed, privatization funds that allow directly to reduce the debt, and indirectly to improve the fiscal budget due to lower interest payments.

Finally, and not surprisingly, privatization is more likely where large and liquid stock markets are in place. The coefficients of the (lagged) market capitalization and the turnover ratio are positive and statistically significant. Well-developed financial markets are key as they allow the absorption of big share issues. Liquidity is also particularly important as after market liquidity is discounted in privatization prices, allowing governments to raise more proceeds. The turnover ratio is also a measure of market activity, which typically increases with a bull market. The positive sign of the coefficient can also be interpreted as governments taking advantage of "hot markets" to float companies, in order to fetch a better price.

Apart from macroeconomic factors, the paper takes into account political and institutional factors as possible determinants of privatization. Particularly, the authors develop an index which includes three components (*i*) a measure of the disproportionality of the electoral rule (the Gallagher index); (*ii*) the effective number of parties; (*iii*) an indicator of the type of executive. Higher values of the political institutional index are associated with a better fit with the majoritarian model, while lower values with the consensus model, which is characterized by strong veto players and a more disperse-decision making power (Lijphart, 1999).

As predicted by the theoretical literature on the political economy of stabilization policies (Alesina and Drazen, 1991; Spolaore, 2003), majoritarian countries privatize more: where less power is granted by the electoral rule to minorities, minority will is underrepresented, and a lower number of veto players is found in the political arena, Therefore large scale reforms packages (which typically include privatizations) are more likely to be implemented. The feasibility of a large scale privatization program therefore is affected by institutional constraints.

5. The outcomes of privatization processes

In this section we will try to investigate some of the main consequences of privatization in Europe, both at the macro and microeconomic level. At the macroeconomic level, we will try to shed some light on the effects of privatization on (i) fiscal conditions of European countries, and (ii) on financial market development. At the microeconomic level, we will try (i) to survey the evidence on the performance improvements of privatized SOEs; (ii) to document the role of share issue privatizations in fostering popular capitalism and ownership diffusion; and finally (iii) to understand whether privatization involved substantial changes in corporate governance structures of European privatized companies.

5.1 The impact on macroeconomic variables

To our knowledge, the effects of privatization on macroeconomic variables have never been the object of a formal analysis. The empirical literature addressing these issues is also quite limited, and no solid evidence has yet been produced in the context of western European countries. This is rather surprising because macroeconomic conditions, especially high deficits and public debt, have been key driver in the decision to privatize the European SOE sector.

In principle, the purpose of privatization is to achieve a redeployment of assets from the public to the private sector. Under the assumption that assets are used inefficiently by the public sector, privatization should spur productivity and the growth of aggregate output. Given that it may take some time for these effect to materialize, one could claim that a sustained privatization policy should foster long term growth and increase output levels of the economy.

Apart from long term considerations, as noted by Mackenzie (1998), privatization programs may have important macroeconomic consequences also in the short run, and especially on public finance aggregates.

To evaluate these effects, it is important first to establish the valuation differential of the assets in public vs private hands, namely the difference between the present value of the income streams generated under public and private ownership. Second, an assumption should be made about the allocation of privatization proceeds.

Suppose that the valuation differential is negative and that revenues are used to retire outstanding debt. Then, the operation improves the public sector net worth and, thanks to lower interests payment loosens the government inter-temporal budget constraint. Private sector wealth and consumption should not fall, since due to the operation the private sector reduces its holdings of money, and increases its holdings of less liquid financial assets. Privatization may instead affect private investment, in the sense that it could crowd out investment that would be otherwise undertaken. Aggregate demand could therefore fall unless the government sterilizes this effect by using the proceeds to finance new investments.

Suppose instead that the valuation differential is positive. This may stem from a different degree of risk aversion: public assets are discounted at higher rates by the private sector rather than by the government, and privatized assets are transferred at underpriced values. In this case, the operation worsens the public sector net worth and tightens the inter-temporal budget constraint since future income flows are not reflected in privatization prices. Under this circumstance there may be a limited effect on investment, but a sizeable wealth effect if the private sector perceives the windfall gain. Privatization policy may therefore have an expansionary impact in the short run.

Suppose instead that proceeds are treated as fiscal revenues (and so put "above the line") and used to finance budget deficits. In this case, the public sector net worth would decrease (independently from the valuation), as the government sells fixed assets to finance public expenditure or tax cuts.

The final effects of privatization on public finances depend also on the possible reduction of State subsides to SOEs, the increase on tax revenues due to the increased profitability of privatized firms, and importantly on the consequences of privatization in the labor market.

Indeed, privatization may involve lay-offs and labor shedding in SOEs, with a possible increase in unemployment benefits in the short run. In the long run it is likely that the positive effects of privatization in terms of increased productivity may spur economic growth and bring the economy to a lower equilibrium level of unemployment.

The overall macroeconomic effects of privatization have never been properly addressed in a general theoretical model which could provide testable predictions. However, a consensus view is that *ceteris paribus* privatization improves fiscal conditions when privatization proceeds are treated as financing and not as a source of budgetary revenue (Lopez Calva and Sheshinski, 2000).

Some empirical work has been carried out to test the effect of privatization on fiscal deficit in Spain, Greece, Italy and Portugal using data for the 1990-1997 period. Jeronimo, Pagán, and Gökçe (2000) find a negative and statistically significant relationship between privatization revenues and deficits for the 1990–1997. Katsoulakos and Likoyanni (2002) examine the impact of privatization on public deficit, on public debt and on other macroeconomic variables (employment and growth) using country level panel data of 23 OECD countries for the period 1990 – 2000.

Katsoulacos and Likoyanni do not find any correlation of privatization variables with budget deficit neither for the whole OECD sample, nor for the four southern European countries. Thus the results by Jeronimo, Pagán, and Gökçe (2000) do not seem to be robust to an extension of the sample period (from 1997 to 2000). However, Katsoulacos and Likoyanni document a statistically significant and negative relation between privatization revenues and public debt. Current privatization receipts have a statistically significant and negative effect on the current

14

unemployment rate and a positive effect on previous period's unemployment rate. When privatization is announced, the announcement is typically followed by restructuring and lay-offs causing an increase in the unemployment rate. When privatization is instead implemented, output may grow, increasing the demand for labor and, thus, decreasing unemployment rate. Finally, Katsoulacos and Likoyanni find, rather surprisingly, that the relations between GDP growth and current or past period's privatization receipts are statistically insignificant for the whole OECD sample.

These results suggest that privatization may be strongly correlated to macroeconomic variables, but certainly more theoretical and empirical work is needed to understand more fully the channels through which privatization affects aggregates.

5.2 Financial market development

The development of equity markets has been one of the main objectives of divestiture throughout Europe. The British privatization program certainly represents one of the most successful experiences. However, the first experiment to strengthen equity culture through privatization in recent financial history was carried out during the 1960s in Germany by the Adenauer government. Subsequent and more ambitious programs to jumpstart or revitalize national exchanges are reported in France in the 1980s under Chirac government, and especially in Italy throughout the 1990s.

But beyond national programs and announcements, it is important to examine first whether privatization is consistent with the objective of stock market development, and second how European governments designed sales to achieve it.

Obviously, as the US experience clearly shows, stock markets could flourish without privatization. Furthermore, other policies, such as the reduction of tax rates on dividends, the establishment of efficient trading infrastructure, and the enactment of sound financial regulation are likely to promote financial market development. Yet a sustained privatization process based on the floatation of shares of SOEs in the stock market should have a strong effect in jumpstarting a market caught in a "low liquidity" trap.

The theoretical models backing this prediction are Pagano (1993), and Subramanian and Titman (1999). The basic assumption is that the listing decision involves important positive externalities on other market participants. But when a company goes public, the entrepreneur fails to recognize the beneficial effect in terms of improved diversification opportunities stemming from his decision. Hence, financial markets can be trapped in a bad equilibrium with few listed firms and high risk premia. The government, as the single owner of several companies, can shift away the market from this bad state through a sequence of IPOs and secondary offerings of privatized companies, and hence reduce the risk premium, and improve overall market liquidity.

A second important aspect is related to the sheer size of SOEs. These companies are usually the largest firms in the country, and even a partial floatation may have a large effect on market capitalization and free float. Importantly, the limited absorption capacity of national stock markets have often induced government to tap major foreign exchanges by cross-listing shares at home and abroad. This privatization strategy stimulates the participation of foreign investors and may reduce the risk premium. Even though it concerns primarily the cross-listed firms, foreign participation will also benefit the liquidity of shares traded only in the local market. If the returns of privatized and local companies are positively correlated, foreigners will share some of the risk borne only by domestic investors prior to privatization. This reduces the required risk premium and thereby increases the value of domestic shares (Chiesa and Nicodano, 2003).

There are sound theoretical reasons to claim that share issue privatization, i.e. the privatization on public equity market as opposed to private placements to strategic investors, promotes financial market development. It is now important to document whether or not the choice of the privatization method is consistent with this stated objective. One would expect privatization on public equity markets to be implemented more frequently in countries where governments are more eager to boost domestic financial development. Obviously this choice involves a trade-off. Share issue privatizations are likely to fail in fledgling stock markets for the simple reason that it is more difficult to find buyers, and that offerings have to be more strongly underpriced (Dewenter and Malatesta, 1997). Due to the costs of using the public capital markets, governments may opt for private sales in less developed capital markets.

To our knowledge, this trade-off has never been the object of empirical analysis in the European context. However, Megginson, Nash, Netter and Poulsen (2000) study the choice of the privatization method for a large sample of (mainly developed) countries, finding that the objective of financial market development dominates – on average – revenue maximization. Indeed, share issue privatizations are more likely in countries with a lower turnover ratio, even controlling for public finance conditions via budget deficits.

Overall, these results suggest that financial market development matters in the choice of the privatization method. But have European governments been able to achieve it ?

A bulk of evidence can be set forth to document the dramatic change in the European financial landscape in the last two decades, when large scale privatization processes were in progress.

As table 3 shows, the average total market capitalization relative to GDP in Europe has increased five times from 1985 to 2000, from 18 to 91 percent. Finland boasts the largest leap throughout the period, from less then 10 to more the 200 percent of GDP. Financial market development has also been remarkable in France, Italy and Spain, where the market capitalization has increased even more then the European average. It is not obvious to quantify the impact of privatization on market capitalization, given the presence of indirect effects via spill-overs and cross-assets externalities. However, a first indication can be provided by looking at the change in the weight of privatized companies on total market capitalization. On average, the relative contribution of privatized companies has increased from 14 to 34 percent, which can partly be ascribed to privatization IPOs, but also the enhancement of market value of newly privatized firms. Interestingly, countries such as France, Italy and Spain report the highest increases in the relative weight of privatized companies together with above-the-average increases in market capitalization. Privatization activity may have therefore played an important role in deepening European stock markets.

[INSERT ABOUT HERE Table 3]

Some interesting facts are also found by looking at the evolution of trading activity measured by the turnover ratio (the total value of trades relative to total market capitalization). On average the turnover ratio increased about four times in the 1985-2000 period. The development of trading volume has been particularly marked in countries such as Finland, Spain, Portugal, and Sweden. The fraction of trading in

shares of privatized companies has dramatically increased over this period, from a bare 7 percent in 1985 to 32 percent in 2000. Interestingly, countries such as Italy, Spain and Portugal, where trading in privatized stocks increased the most, report also substantial variation in total trading activity, a preliminary finding which suggests that privatization may foster financial market development above the mechanic increase in market capitalization.

However, market capitalization and trading volumes fail to capture a fundamental aspect of financial market development, namely *liquidity*. Liquidity is important because it allows companies to raise capital more cheaply (Ellul and Pagano ,2004), and to design stock based managerial incentives schemes (Holmstrom and Tirole 1993), spurring company performance, efficiency, and ultimately economic growth. But liquidity is a quite elusive concept which is hard to define, let alone to quantify. The bid-ask spread has emerged as a conventional proxy for liquidity but these spreads are difficult to compare across countries due to differences in market microstructure. Some of these difficulties can be circumvented by the use of price impact measures, given by the absolute value of return scaled by volume traded.

Bortolotti, De Jong, Nicodano and Schindele (2004) estimate the effect of share issue privatization on stock market liquidity measured by the price impact in OECD countries, while accounting for other potential determinants set forth in the literature, such as the enforcement of insider trading regulation, political and country risk, and capital markets liberalization. They find that privatization represents a source of variation of market liquidity. Particularly, as predicted by theory, the international profile of privatization matters the most. The price impact (hence liquidity) appears to be strongly and negatively (positively) correlated with the quantity of shares allocated to foreign investors in international exchanges. Importantly, the effect of cross-listings at the privatization stage survives when the liquidity of *private* companies is considered. A large scale privatization program based on international SIPs generates important positive externalities on the liquidity of private companies as well by improving diversification opportunities and by reducing risk premia.

5.3 The financial and operating performance of European privatized firms

There is no study available in the literature investigating the performance of European privatized firms in a single and comprehensive statistical analysis. However, some information could be grasped from Megginson, Robert Nash and Matthias van Randenborgh (1994) and Juliet D'Souza and Megginson (1999) (MNRD) papers, dealing with the pre-post privatization performance changes in relatively large samples of privatized companies. The majority of these 133 companies are from industrialized economies, with a large predominance of European countries, so that the information coming from these papers – with a lot of caveats – could represent a useful starting point. We will then try to complement the analysis of performance reporting results of some country studies in the European context.

MNRD studies yield consistently positive results on the effectiveness of privatization in promoting improvements in the financial and operating performance of divested companies. By documenting economically and statistically significant post-privatization increases in real sales (output), profitability, efficiency (sales per employee), and capital spending, coupled with significant declines in leverage. This point is made clear in Table 4, which summarizes the results of the two studies.

[INSERT ABOUT HERE Table 4]

Additionally, these two studies consistently document that output, efficiency, and capital spending increase dramatically and significantly after privatization. Meanwhile, leverage declines significantly. Megginson (2003) comments these results concluding that "unlike profitability increases, these are all unambiguously socially beneficial outcomes, since they imply that privatized firms use resources more productively and also become financially healthier. That these benefits are achieved without systematically reducing employment also suggests that privatization yields important social benefits. In sum, the weight of evidence in these studies clearly indicates: (1) that privatization improves the operating and financial performance of newly divested firms, (2) that these improvements are the result of socially beneficial improvements in productive efficiency and entrepreneurial effort, and (3) that privatization "works" in a wide variety of countries, industries, and competitive environments."

The methodologies pioneered by MNRD have become standard in the empirical analysis of the performance of privatized firms. However, they suffer from several drawbacks. First, selection bias raises probably the most serious concern, as the sample is made of companies sold in public equity markets via share issue privatization (SIP) programs. These companies tend to be the largest and usually the most profitable SOEs, which due to intense restructuring pre-privatization are certainly the easiest to privatize. Second, the two snapshots taken on performance measures pre and post privatization do not allow disentangle the possible sources of these improvements, which may be ascribed to privatization *per se*, but also to other factors such as a lack of competition or weak regulation.

Some steps in this direction have been made by Bortolotti, D'Souza, Fantini, and Megginson (2002) in a study on the global telecommunication industry, virtually including all major European operators. Using panel data models, it is found that some performance measures are more strongly affected by competitive conditions (with higher profitability associated with less intense competition in the product markets) rather than the privatization alone.

The existing evidence stemming from cross-country analyses does not allow to conclude that privatization *per se* has been the key in boosting the financial and operating performance of firms, but rather the combination of liberalization, regulatory and ownership changes.

In what follows, we will try to complement this evidence by summarizing the main results of country studies on the economic and financial effects of privatization on the behavior of privatized SOEs in European context, starting from the UK experience.

Several studies document significant performance improvements along some key performance measures but most of them conclude that the program could and should have been executed with more concern for distributional issues and/or with greater protection built for consumers.

Saal and Parker (2003) examine the productivity and price performance of the privatized water and sewerage companies of England and Wales after the industry was privatized and a new regulatory regime imposed in 1989. They document that labor productivity improved significantly after privatization, but they find no evidence that total factor productivity grew as a direct result of the ownership change. They also find that increases in output prices have outstripped increased input prices, leading to significantly higher economic profits after privatization.

Newbery and Pollitt (1997) perform a counter-factual analysis of the 1990 restructuring and privatization of the UK's Central Electricity Generating Board

(CEGB), and document significant post-privatization performance improvements. However, they find that the producers and their shareholders capture all of the financial rewards of this improvement and more, whereas the government and consumers lose out.

Price and Weyman-Jones (1996) measure the technical efficiency of the UK natural gas industry before and after its 1986 privatization and associated regulatory changes. They employ non-parametric frontier analysis to show that the industry's rate of productivity growth increased significantly after privatization - though not as much as it could have if the industry had been restructured and subjected to direct competition and more appropriate regulation.

In a comprehensive case-study on the UK, Florio (2004) uses cost-benefit analyses to investigate the effect of privatization on firms, consumers, shareholders, workers, and taxpayers, concluding that the overall effect of "the Great Divestiture" on efficiency has been modest, and that privatization had a substantial regressive effect on the distribution of incomes.

Studies conducted in other European countries provide very mixed evidence. Villalonga (2000) examines the effect of privatization on the operating efficiency of 24 Spanish firms that were fully divested between 1985 and 1993. Privatization seems to decrease efficiency over the intermediate term (5 and 6 years after divestiture), but to increase efficiency over the longer term (7 and 8 years) afterwards and in the period leading up to privatization (4 and 3 years before).

Dumontier and Laurin (2002), investigate the value that was created or lost during the State ownership period for each of the 46 French companies (39 banks and 5 industrial firms) that were nationalized during 1982 and then re-privatized between 1986 and 1995. They analyze whether the subsequent privatization of these companies improved performance over that achieved during the post-1982 nationalized period. They find that the French government created value in the nationalized firms, but the State and taxpayers did not benefit because of the premium that was paid to shareholders upon nationalization (20 percent), and because of the underpricing of the IPOs at the time of privatization. The financial and operating performance of companies improved during the nationalization phase, then improved even more after privatization. Employment fell during the nationalization, but increased (due to higher sales) after privatization. Goldstein and Nicoletti (2003) conduct a performance analysis based on a sample of 25 Italian privatized non-financial firms. Their analysis is important as it is one of the few based on a comparison with a control group of private companies, which allows to take into account cyclical movements in the economy. Interestingly, none of the traditional indicators used in MNRD studies gains statistical significance, with the exception of investment indicators which show a marked improvement.

The evidence presented in this section does not allow us to provide an unambiguous answer about the role of privatization on the financial and operating performance of European SOEs. First, the bulk of the evidence coming from crosscountry studies does not appear to be robust. Second, countries studies provide mixed results. Importantly, the implication of privatizations in terms of allocative and productive efficiency have not yet empirically been documented in comprehensive statistical analyses.

5.4 Popular capitalism and ownership diffusion

The political economy approach to privatization points out that fostering popular capitalism and widening share ownership are possible objectives of divestiture which right wing market-oriented governments may find particularly attractive. The reason for this preference is not purely "ideological", but grounded on self-interest and political opportunism. Indeed privatization, by making equity investment attractive for the middle classes, can create a constituency with an interest in increasing the value of its assets and therefore averse to the redistribution policies of the left. In this way, privatization can be a rational strategy for raising the probability of electoral success of market-oriented coalitions. The key variable to achieve re-election via privatization is underpricing, which in turn depends on income distribution. Indeed, the poorer is the median voter, the more underpricing is needed to entice him to become a shareholder in privatized firms (Biais and Perotti, 2002).

The English experience seems to fit in quite well with the empirical implication of this model. The Thatcher government's privatization program, especially in the initial term, was implemented with the declared objective of expanding and spreading equity ownership. This was achieved through a massive program of share issue privatizations (SIPs) characterized by substantial underpricing.⁴ In this way, the distribution of equity at a discounted price made the re-nationalization (proposed in

the Labour party's electoral program) costly while simultaneously increasing conservative support. Indeed, the five consecutive victories by the Conservatives indicate that strategic privatization (in combination with other market oriented policies such as tax relieves, reduction in public expenditures, and deregulation) may have paid off at general elections.

Beyond the English experience, it is commonly recognized that privatization are politically motivated. A bulk of empirical papers have provided evidence that partisan politics matters in the choice of the privatization method. Particularly, privatization implemented by right wing governments in developed (mainly European) countries tend to be structured as public offers instead of private equity placements (Bortolotti and Siniscalco, 2004; Bortolotti and Pinotti, 2004), to be more strongly underpriced where the income inequality is higher (Jones, Megginson, Nash, and Netter, 1999), and to exhibit a preferential allocation of shares towards domestic retail investors (Bortolotti and Siniscalco, 2004).

The available evidence appears broadly consistent with the idea that political objectives shape privatization. However, the empirical literature has not established whether privatization contributed to widen share ownership, and to promote popular capitalism, nor that it has significantly shifted political preferences by creating support to market oriented policies in Europe.

Boutchkova and Megginson (2000) analyze the evolution of share ownership in a sample of SIPs, concluding that the initial structure of shareholding does not appear to be stable over the long run. Indeed, the striking number of initial shareholders (often over 100,000) declines by 33 percent within five years of the offering. The privatization process in the United Kingdom seems to point this direction, as the inflation in the number of shareholders in privatized firms has been a temporary phenomenon. Clarke and Pitelis (1993) document substantial individual flipping such that the majority of initial investors immediately disposed of their holdings to cash the initial discount. The majority of shares ended up eventually with the financial institutions.

In a comprehensive analysis of retail incentives in share issue privatizations from 1981 to 2003, Kelohariju, Knupfer, and Torstila (2004) come to a sharply different conclusion with respect to the UK case-study. First, they document a widespread use of retail incentives such as bonus shares, i.e. a free distribution of shares to investors

⁴ See Biais and Perotti (2002); Jenkinson and Ljunqvist (2000).

holding shares for a given period (frequently 24-36 months). Second, they document that these measure are costly, but extremely effective in attracting retail investors. Particularly, they find that a dollar spent in retail incentives increases the number of investors participating in the offering 21 times more than a dollar spent on underpricing. Finally, in an interesting controlled experiment based on Finnish privatizations, they document that flipping is not simply postponed at the end of the lock-up period but substantially reduced in bonus tranches as compared to regular tranches.

Retail incentives (which have been regularly used in share issue privatization) have worked well to meet the goal of widening the domestic shareholder base. The fact that these incentives are absent in private sector offerings confirms (again) that political objectives shape divestiture. It would be interesting to know more about the role of political preferences towards these incentives. Should bonus shares and incentives be systematically associated with share issue privatization by right wing governments, then a final test could be provided about the link between partisan politics and the objective of spreading share ownership and equity culture.

5.5 Ownership and control in privatized firms

The wave of privatization occurred in Europe during the 1990s definitively represents one of the greatest transfer of ownership in the history of the corporation. However, there is a lingering belief that privatization did not alter dramatically the corporate governance of SOE, which governments still hang on to control by direct and indirect means.

This section presents updated empirical evidence on the transfer of ownership and control in European privatized companies. The transfer of ownership is measured by the stake sold in the various operations in different countries. The transfer of control, instead, is analyzed through the government's voting rights in privatized companies, and by the temporary or permanent restrictions to the control rights of the private investors such as "golden shares".

The divestiture of minority holdings, or partial privatization, appears to be quite common in Western Europe. From 1977 to 2003, 59 percent of the 1,133 deals involved the sale of the majority of stock. Interestingly, this percentage shrinks to 21 percent in the sample of privatizations through public offer, where the average percentage of capital sold is 36 percent.

Table 5 shows the average percentage of capital sold in privatization deals in European countries. Obviously, a distinction has to be made between public offerings, which involve the largest companies and often more profitable SOEs floated in the stock market, and private placements, which instead are typically used to sell small sized firms operating in non strategic sectors. The data show that privatization of the latter tends to be more complete especially in countries such as the United Kingdom, Sweden, Germany, France and Finland. On the contrary, partial privatization is typical of larger SOEs, which are usually sold by tranches. Indeed, lower averages of capital are sold trough public offers of shares in almost all major European countries, with the exception of the United Kingdom.

[INSERT ABOUT HERE Table 5]

Partial privatization is certainly an interesting feature of State's assets disposal. The effect of partial privatization should not be understated, given that the initial listing of the shares of the SOE can have first order effects on managerial incentives and performance (Gupta, 2004). However it is important to document whether partial sales are just a snapshot of a process that will end with a complete divestiture, nor they tend to persist over the long run, as if governments were not really intended to give up ownership and control.

While privatizing the first tranche represents a win-win solution, allowing the government to raise revenues and to enhance the company value, the complete relinquishment of control is a politically costly decision given that governments lose a powerful instrument for targeted redistribution (as high wages and job security could be earmarked to special categories of workers), the right of having representatives in the boards in order to affect corporate decisions, and the power to safeguard public interests and national security.

Bortolotti and Faccio (2004) dig into the issue of government's control by analyzing the recent evolution of ultimate voting rights in 141 privatized companies, of which 81 percent from Western Europe.

Ultimate (direct and indirect) control (voting) rights by private and public shareholders (including the central State, federal or regional bodies, Central Banks,

etc.) are computed as the weakest link along the control chain, taking into account pyramiding and cross-holdings (as in La Porta, Lopez de Silanes, Shleifer, and Vishny 1999). Then several categories of privatized firms are identified according to the identity of the largest ultimate controlling shareholder at the 10 percent cut-off level (see Table 6).

[INSERT ABOUT HERE Table 6]

In fact, the privatization process in developed economies (and particularly in Europe) does not seem to be accomplished. As of 2000, the State is *still* the largest shareholder in almost 30 percent of the privatized firms of our sample. The rest of the sample is split between family controlled (19 percent) and widely held firms (30 percent).

The high percentages of State-controlled firms observed in year 2000 feed the suspicion that privatization has been carried out reluctantly during the 1990s. In order to quantitatively assess this reluctance, we take a second snapshot in a previous year.

Data availability allows us to go back in time to 1996, when we find a higher number of State-controlled firms (34 percent), a lower number of widely held (27 percent) and privately owned companies (16 percent). During the period, the government has relinquished control only in 10 companies, which account for 7 percent of our sample.

The analysis of ultimate voting rights of the largest shareholders reported in Table 7 yields some interesting results. In 2000, the largest shareholder on average controls 25 percent of voting rights. When the government is the largest shareholder, it controls more than 50 percent of voting rights. "Government-controlled" privatized firms show therefore a much higher concentration of power, which does not seem to decline over time.

[INSERT ABOUT HERE Table 7]

We now beg a final question. Is it possible to identify economic reasons to rationalize governments reluctance to sell, or has it to be traced back to their willingness to keep companies under political control? More precisely, are there idiosyncratic factors - maybe related to a given country, sector, or business activity - which could explain why some SOEs are so tightly controlled?

One possible way of testing this hypothesis is to construct a control sample of private firms, and then to compare the evolution of the ultimate ownership within the two samples.

Table 7 reports some statistics on the ultimate ownership of privatized companies as opposed to their respective matching private firms.⁵ The pooled data suggest a quite strong convergence between the privatized firms and the control sample. The samples report a statistically significant difference in means of approximately 7 percent in 1996, which becomes insignificant and negligible in 2000.

In the sub-sample of government controlled firms, control structures do not converge at all. On average, in 1996 the public shareholder owns a control stake which is 30 percent higher than the one owned by the largest ultimate shareholder in private firms. This difference shrinks only marginally at the end of the period, while remaining highly statistically significant.

This evidence about the dynamics of ownership in privatized companies allows a deeper understanding of government's reluctance to sell. On average, the European privatization process has contributed to the roll back of the State in the ownership of productive assets. Several companies have been sold off, and in these companies the governments do not appear as the major shareholders post-privatization. However, there is also a hard core of companies which remain tightly controlled by the State over time.

The previous analysis has shown that several State-owned enterprises have been fully privatized during the 1990s. However, the sale of a majority holding is not itself a sufficient condition to avoid government interference in privatized companies. Governments can grant themselves wide discretionary powers over partially or even fully privatized companies by the use of "golden shares". By exerting its rights, the "special" shareholder can often influence the choice of management, exert veto power over the acquisition of relevant stakes by private shareholders, even without owning the majority of stock in the company, or a single share of capital.

Golden Share can broadly be defined as the complex of special powers granted to the State and the statutory constraints in privatized companies. Typically, special powers include (i) the right to appoint members in corporate board; (ii) the right to

⁵ For a detailed description of the methodology used to identify the control group see Bortolotti and Faccio (2004).

express a consent or to veto the acquisition of relevant interests in the privatized companies; *(iii)* other rights such as consent on the transfer of subsidiaries, dissolution of the company, ordinary management, etc. The above mentioned rights may be temporary or not. Statutory constraints instead include *(i)* ownership limits; *(ii)* voting caps; *(iii)* provisions of national control.

As Table 8 shows, in Western European countries golden shares are widespread. In 1996, they are found in more than 42 percent of the companies in our sample, and in each and every country. Furthermore, they are highly concentrated in some sectors, such as defense, where 100 percent of the privatized companies have golden share, telecommunications (83 percent), oil and gas (62 percent), utility (64 percent) and transports (40 percent).

Indeed, governments resort to golden shares in order to protect a broadly defined concept of "national security" as they allow to shielding privatized companies operating in the defense business from hostile (foreign) take-overs but also utilities providing public services, such as gas, electricity, water, telecommunications, and transports. The provision of such services but also the safeguard of essential facilities are certainly strategic, especially when privatization did not proceed in parallel with adequate liberalization and effective regulation.

Table 8 provides a comprehensive account of government power in privatized firms, by taking into account in combination voting rights and golden shares mechanisms. As of 2000, 64.9 percent of the 118 privatized companies for which reliable information is available are either directly controlled by the state or warrant special powers to the public shareholder through additional control devices.

[INSERT ABOUT HERE Table 8]

This combined evidence shows that governments are reluctant to privatize, and this reluctance appears particularly strong in the so-called strategic sectors. This protectionist attitude can certainly be attributed to governments' willingness to maintain firmly political control over one country's largest and most valuable corporations. However, the failure to relinquish control has also to be traced back to the various economic and institutional constraints shaping economic policy. Indeed, genuine privatization is problematic if markets are not competitive, and regulation is weak.

These new results on the corporate governance of privatized firms beg a final question: Does reluctant privatization matter in the valuation of firms?

According to a largely held view, a principal-agent plagues government controlled firms, as the owners (the taxpayers) have different objectives from the bureaucrat or the politician controlling the firm. In these firms, the manager may run the company to achieve political objectives such as keeping redundant workes, and not to maximize profits. When control rights are transferred to the private sector, more emphasis will be placed on efficiency (Vishny and Shleifer, 1994). This theory has a straightforward empirical implication: in privatized firms, government voting rights are negatively related to the market valuation of a company.

Bortolotti and Faccio (2004) test this hypothesis by regressing ultimate voting rights on (adjusted) market-to-book ratios in a large sample of privatized firms, and find a quite surprising result: higher government control rights are *not* negatively discounted in market values. On the contrary, government controlled privatized companies appear on average more valuable than fully privatized firms. Results do not appear to be driven by reverse causality nor by the agency costs of private ownership, and survive when several control variables (including sector dummies) are included.

Indeed, it is possible that governments grant special benefits to privatized companies in which they retain control, regardless of industries. The potential benefits include subsidizing loans, guaranteeing contracts, and shielding companies from competition.

A caveat is in order. This analysis on the effect of government power in firms are performed in a sample of privatized companies listed in the stock market. The reader should not jump to the conclusion that state ownership is superior since the empirical literature has documented major performance improvements in state-owned companies when they are initially privatized. Rather, the previous result points out the existence of a non linear relation between government rights and market valuation, which should be investigated in a richer theoretical setting where the "grabbing hand" hypothesis is complemented with other hypotheses on the behavior of politicians in firms.

6. Conclusions: the outstanding issues

Privatization has certainly been one of the main events of the economic and financial history of the 20th century. Western Europe had a great bearing in such a process, having raised a half of worldwide proceeds. The relevance of Western Europe in the process can be ascribed to several factors, the main one being the abnormal growth of the SOE sector which occurred during the century to foster the industrialization and stabilize the European economies severely hit by adverse shocks.

At the turn of the century, the process abruptly slowed down at the global scale, and especially in Western Europe, the continent that launched the process.

This stylized fact begs an important question. Is privatization in Western Europe a long term trend proceeding in parallel with the advancing of market capitalism, or rather a cycle following the short or medium term fluctuations of economic fundamentals?

The empirical analysis shows that privatization processes are shaped by economic and political determinants. Particularly, they are affected by market conditions, so that large privatization waves are systematically associated with bull stock markets. But budget constraints and political institutions also matter. Weak fiscal conditions and the urge to meet Maastricht criteria have certainly been major drivers in the privatization decision, as financially distressed governments have been more eager to privatize. However, government preferences face institutional constraints so that large privatization programs have been more smoothly implemented in countries endowed with majoritarian political institutions, curbing the veto powers of averse constituencies and entrenched interests.

We can therefore tentatively conclude that the big privatization wave of the 1980s and especially 1990s has mostly been a cyclical phenomenon, where the engine of sales have been booming stock markets, and worsening fiscal conditions. However, within this common trend, the extent to which governments privatize depends upon exogenous political and institutional determinants, which tend to persist in the long run. As to the effects of privatization, more research is needed to understand the profound consequences of divestiture in Western Europe. However, on the basis of the existing evidence, we can tentatively conclude that the effects at the macroeconomic level have been important especially in terms of stock market development.

The microeconomic effects of privatization are instead less visible, both in terms of the performance of privatized firms and corporate governance structures. First, the financial and operating performance has apparently improved in the aftermath of privatization. However, only limited efforts have been displayed to quantify these improvements using private firms as benchmarks, and to isolate the possible sources of these improvements. Indeed, it is not clear whether they stem from the additional monitoring role played by the stock market, from intentional structural reforms, or from the dilution of government ownership.

As to ownership and control in privatized companies, the empirical literature has provided solid evidence that privatization in Western Europe has been partial and incomplete. In most cases privatization it did not entail a dramatic change in governance structures as private ownership and public control actually seem to coexist.

The final question that we raise is the following: is the coexistence of private ownership with public control just a transient European anomaly, or a functional pattern of governance?

Providing tentative answers to this question is certainly an important avenue for future research, both on the theoretical and empirical side. For the time being, we conjecture that genuine privatization (i.e. the transfer of ownership and control to the private sector) appears difficult to achieve and sustain as several conditions must be met. First, markets should be competitive or suitably regulated. Second, private investors should be adequately protected by the law in order to avoid expropriation. Third, political institutions should be designed to limit the veto power of constituencies ousting full divestiture. Last but not least, governments should be credibly committed not to interfere post-privatization in the operating activity of the companies. Finally, and more cynically, the financial incentives for full divestiture may be limited if governments stakes in privatized firms turn out to be valuable for shareholders, since special benefits could be targeted to (partially) privatized firms.

The most recent trends document a strong resurgence of privatization activity in Europe, especially through public offering of shares to retail investors. European governments, severely hit by the global economic crisis and envisaging fiscal operations to improve budgetary figures, have once again resorted to privatization sales. The positive outlook in stock markets and investor's appetite towards high quality stocks have certainly been key drivers in the resumption of the process. It is certainly hard to predict whether this new trend represents the beginning of a new cycle or simply a re-adjustment after a negative shock. Certainly, the stage of increasing returns of privatization is probably over, and this makes privatizing the "second tranche" a difficult challenge for European governments.

	<u>Privatiz</u>	ation Trans	actions	Privatization Revenues			
Area	Total Deals	%	PO/ Deals	Total Revenues	%	PO/ Revenues	
Western Europe	1,183	0.29	0.36	647,647.75	0.48	0.73	
Asia	569	0.14	0.53	322,349.60	0.24	0.85	
Latin America	501	0.12	0.14	154,499.00	0.11	0.21	
Oceania	200	0.05	0.08	88,237.78	0.07	0.36	
CEECs & Former Soviet Union	1,145	0.28	0.12	84,471,25	0.06	0.25	
North America & Caribbean	115	0.03	0.21	24,187.56	0.02	0.57	
MENA	185	0.05	0.61	20,767.39	0.02	0.64	
Sub Saharan Africa	156	0.04	0.30	7,9300.07	0.01	0.33	
Total	4,054	1.00	0.28	1,350,090.40	1.00	0.64	

Table 1. Privatizations around the world: revenues and transactions by geographic areas (1977-2004)

Source: Elaboration on Securities Data Corporation.

Note: In our classification *Western Europe* includes: Austria, Belgium, Denmark, Finland, France, Germany, Greece, Iceland, Ireland, Italy, Luxembourg, Malta, Monaco, the Netherlands, Norway, Portugal, Spain, Sweden, Switzerland, Turkey and the United Kingdom; *North America and the Caribbean* include: Barbados, Belize, British Virgin, Canada, Dominican Republic, Haiti, Honduras, Jamaica, Netherlands Antilles, Nicaragua, St Lucia, Trinidad, Tobago and the United States; *Latin America* includes: Argentina, Bolivia, Brazil, Chile, Columbia, Ecuador, El Salvador, Guatemala, Guyana, Mexico, Panama, Paraguay, Peru, Uruguay and Venezuela; *Sub Saharan Africa* includes: Benin, Cameroon, Chad, Congo, Ethiopia, Gabon, Ghana, Guinea, the Ivory Coast, Kenya, Lesotho, Malawi, Mali, Mauritius, Mozambique, Nigeria, Rwanda, Sao Tome, Senegal, Sierra Leone, South Africa, Sudan, Tanzania, Uganda, Zambia, Zimbabwe; *the Middle East and North Africa (MENA)* include: Algeria, Bahrain, Egypt, Israel, Jordan, Kuwait, Lebanon, Mauritania, Morocco, Oman, Qatar, Tunisia; *Asia* includes: Armenia, Bangladesh, Cambodia, China, India, Indonesia, Japan, South Korea, Malaysia, Pakistan, the Philippines, Singapore, Sri Lanka, Taiwan, Thailand; *Oceania* includes: Australia, Fiji, French Polynesia, New Zealand, Papua New Guinea; *Central-Eastern Europe and the Former Soviet Union* include: Albania, Bosnia, Bulgaria, Croatia, the Czech Republic, Czechoslovakia, East Germany, Estonia, Georgia, Hungary, Kazakhstan, Latvia, Lithuania, Macedonia, Moldova, Poland, Romania, Russian Federation, Slovak Republic, Slovenia, Soviet Union, Ukraine, Uzbekistan, Yugoslavia.

Table 2. Privatizations in Western Europe (1977-2002)

This table reports the aggregate figures on privatization in Western European countries for the 1977-2002 period. Countries are ranked by *Rev/GDP*. *Deals* is the total number of privatizations; Revenues is Total Revenues from Privatizations for the period 1977-2002 (in US\$ mil 1995); *Po/Deals* is the ratio of the number of privatizations by Public Offer to the total number of privatizations; PO/Rev is the ratio of revenues raised through Public Offers of Shares to Total Revenues from privatizations; *Rev/GDP* is the ratio of total revenues cumulated in the period to 2002GDP (in US\$ mil 1995); SOE/GDP is the ratio of the SOE value added to GDP (in US\$ mil 1995) reported the year before the first privatization; Rev/SOE is the ratio of Revenues (in US\$ mil 1995) reported the year before the first privatization, when possible.

Country	Deals	Revenues	PO/ Deals	PO/ Rev	Revenues/ Deals	Rev/GDP	SOE/GDP	Rev/SOE
	70	05 450 45	0.51			0.10	0.45	4.07
Portugal	/8	25,453.65	0.51	0.80	326.33	0.19	0.15	1.87
United Kingdom	183	145.531.73	0.32	0.88	795.26	0.11	0.11	1.40
Finland	56	16,328.63	0.43	0.66	291.58	0.10		
Italy	103	96,442.39	0.44	0.84	936.33	0.08	0.09	0.86
Spain	74	46,577.60	0.35	0.79	629.43	0.06	0.09	1.01
Sweden	56	18,625.54	0.20	0.70	332.60	0.06	0.10	0.73
Austria	51	11,503.06	0.57	0.51	225.55	0.04	0.14	0.52
Netherlands	29	19,182.48	0.38	0.66	661.46	0.04	0.06	1.15
France	97	59,875.26	0.53	0.92	617.27	0.03	0.11	0.48
Germany	150	73,302.53	0.14	0.66	488.68	0.03	0.10	0.79
Belgium	10	5,707.97	0.20	0.18	570.80	0.02	0.04	0.69
Mean	81	47.139.17	0.37	0.69	534.12	0.07	0.09	0.95
Median	74	25.453.65	0.38	0.70	570.80	0.05	0.10	0.82

Source: Elaboration on Securities Data Corporation, World Development Indicators (World Bank 2002), and World Bank 1995.

Table 3. Financial Markets Development Indicators (1985, 2000)

	Market Capitalization							Volume of Trades			
	1985	2000	1985	2000	D ://	1985	2000	1985	2000		
	Mkt. Cap (as % of GDP)	Mkt. Cap (as % of GDP)	Mkt. Cap. of Privatized Companies as a % of Total Mkt. Cap (a)	Mkt. Cap. of Privatized Companies as a % of Total Mkt. Cap (b)	Difference (a) – (b)	Turnover Ratio	Turnover Ratio	Value of Trades of Privatized Companies as % of Total Volume of Trades (c)	Value of Trades of Privatized Companies as % of Total Volume of Trades (d)	25 (c) -(d) 26 26 27 27 28 29 29 29 29 20 20 20 20 20 20 20 20 20 20 20 20 20	
Austria	5.00	15.00	2.00	34.00	32.00	23.00	33.00	1.00	49.00	47.00	
Belgium	0.00	1.00	1.00	8.00	8.00	11.00	23.00	1.00	11.00	10.00	
Finland	9.00	236.00	1.00	14.00	12.00	10.00	72.00	2.00	20.00	18.00	
France	11.00	99.00	26.00	83.00	57.00	25.00	84.00	3.00	35.00	32.00	
Germany	18.00	64.00	4.00	22.00	18.00	55.00	89.00	4.00	15.00	11.00	
Italy	10.00	62.00	17.00	41.00	24.00	33.00	117.00	2.00	34.00	32.00	
Portugal	16.00	53.00	36.00	57.00	21.00	17.00	96.00	19.00	61.00	42.00	
Spain	14.00	74.00	34.00	56.00	21.00	44.00	238.00	30.00	66.00	36.00	
Sweden	30.00	136.00	1.00	6.00	6.00	31.00	125.00	3.00	18.00	15.00	
United Kingdom	63.00	171.00	15.00	16.00	1.00	24.00	75.00	7.00	11.00	4.00	
Average	17.60	91.10	13.70	33.70	20.00	27.30	95.20	7.20	32.00	24.70	

Source: Elaboration on Word Bank (2003), and Datastream.

Table 4. Results from three empirical studies on the financial and operating performance of newly-privatized firms

This table summarizes the empirical results of three directly-comparable academic studies [Megginson, Nash and van Randenborgh (1994), and D'Souza and Megginson (1999)] comparing the three-year average operating and financial performance of a combined sample of 133 newly-privatized firms with the average performance of those same firms during their last three years as state-owned enterprises (SOEs). All three studies employ the Wilcoxon rank sum test (with its z-statistic) as the test of significance for the change in median value. All three studies employ multiple proxies for most of the economic variables being measured; this table summarizes only one proxy per topic, and emphasizes the one highlighted in the studies (almost invariably, the variable that uses either physical measures--such as number of employees--or financial ratios using current-dollar measures in the numerator or denominator, or both). Profitability, investment, leverage, and dividend measures are in percent. Efficiency and output measures are index values, with the value during the year of privatization defined as 1.000; inflation-adjusted sales figures are used in the efficiency and output measures.

Variables and Studies cited	Obs	Mean value Before Privatization	Mean value After Privatization	Mean change due to Privatization	Z-Statistic for Difference in Performance	% of Firms with improved of Performance	Z-Statistic % change
PROFITABILITY (Net							
Income÷Sales) Megginson, Nash and	55	0.0552	0.0799	0.0249	3.15***	69.1	3.06***
van Randenborgh (1994) D'Souza &	78	(0.0442) 0.14	(0.0611) 0.17	(0.0140) 0.03	3.92***	71	4.17***
Megginson (1999)		(0.05)	(0.08)	(0.03)			
EFFICIENCY (Real Sales per Employee)							
Megginson, Nash and	51	0.956 (0.942)	1.062 (1.055)	0.1064 (0.1157)	3.66***	85.7	6.03***
(1994) D'Souza &	63	1.02 (0.87)	1.23 (1.16)	0.21 (0.29)	4.87***	79	5.76***
Megginson (1999)							
INVESTMENT (Capital	43	0.1169	0.1689	0.0521	2 35**	67.4	2 44**
Expenditures Sales) Megginson, Nash and	45	(0.0668)	(0.1221)	(0.0159)	0.80	55	0.81
van Randenborgh (1994) D'Souza &	00	(011)	(0.10)	(-0.01)	0.00	55	0.01
Megginson (1999)							
OUTPUT (Real Sales (adjusted by CPI))							
Megginson, Nash and	57	0.899 (0.890)	1.140 (1.105)	0.241 (0.190)	4.77***	75.4	4.46***
(1994) D'Souza & Megginson (1999)	85	0.93	2.70	1.76	7.30***	88	10.94***
EMPLOYMENT (Total							
Employees) Megginson, Nash and	39	40,850	43,200	2,346	0.96	64.1	1.84*
van Randenborgh (1994) D'Souza &	66	(19,360) 22,941	(23,720) 22,136	-805	-1.62	36	-2.14**
Megginson (1999)		(9,876)	(9,106)	(-770)			
LEVERAGE (Total Debt Total Assets)							
Megginson, Nash and	53	0.6622 (0.7039)	0.6379 (0.6618)	-0.0243 (-0.0234)	-2.41**	71.7	3.51***
(1994) D'Souza & Megginson (1999)	72	0.29	0.23	-0.06	-3.08***	67	3.05***
DIVIDENDS (Cash							
Dividends Sales) Megginson, Nash and	39	0.0128	0.0300	0.0172	4.63***	89.7	8.18***
van Randenborgh (1994) D'Souza &	51	0.015	0.04	0.025	4.98***	79	5.24***
Megginson (1999)							

*** Indicates significance at the 1 percent level

** Indicates significance at the 5 percent level

• Indicates significance at the 10 percent level

Source: Megginson, 2003.

Table 5. Percentages of capital sold in European privatizations (1977-2003)

This table reports aggregate figures on privatization in major Western European countries for the 1977-2003 period. Countries are ranked by Average percentage of Capital Sold. *Deals* is the total number of privatizations.

Country	Deals	Average percentage of capital sold	Average percentage of capital sold through PO	Average percentage of capital sold through PS	
United Kingdom	186	89.92	73.84	96.95	
Germany	156	78.08	29.31	84.99	
Sweden	59	77.44	33.45	87.73	
Ireland	17	65.20	38.35	73.46	
Spain	80	63.65	30.01	79.22	
Netherlands	32	60.64	24.92	82.08	
Norway	34	59.95	34.16	76.07	
France	114	59.26	37.81	75.99	
Portugal	81	55.24	36.02	75.46	
Italy	124	54.71	29.40	70.70	
Finland	58	54.59	17.70	81.43	
Denmark	9	46.14	32.71	68.53	
Austria	58	45.70	32.73	61.77	
Belgium	11	41.88	33.30	43.79	
Mean	73	60.89	34.55	75.58	

Source: Elaboration on Securities Data Corporation.

Table 6. Type of Largest Shareholder in Privatized Firms

Data for 141 privatized firms are used to construct this table. The table presents the percentage of firms controlled by different controlling owners, using 10% ownership as the threshold. Controlling shareholders are classified into six types. State: A national government (domestic or foreign), a local authority (county, municipality, etc.), or a government agency. Family: A family (including an individual) or a firm that is unlisted on any stock exchange. Widely held financial institution: A financial firm (SIC 6000-6999) that is defined as widely held because no shareholder controls 10% or more of the votes; held at the control threshold. Widely held corporation: A non-financial firm, widely held using the control threshold. Crossholdings: The firm Y is controlled by another firm, that is controlled by Y, or directly controls at least 10 percent of its own stocks. Miscellaneous: Charities, voting trusts, employees, cooperatives, or minority foreign investors. Companies that do not have a shareholder controlling at least 10 percent of votes are classified as widely held. ^a, ^b, and ^c denote statistical significance at the .01, .05, and .10 levels, respectively.

	Privatized Firms									
Time period	N° of	State	Family	— of и	hich:	Widely	Widely	Miscell.	Cross-	Widely
	firms			Identified families	Unlisted firms	held corp.	held financial		holdings	held
End of										
1996	141	34.75	16.31	2.84	13.48	2.84	17.02	1.42	0.00	27.66
End of										
2000	141	29.79	19.86	2.84	17.02	4.26	9.93	4.96	0.71	30.50
		t o (b					z oo b	o ee b		
DIII 00-90		-4.96	3.55	0.00	3.55	1.42	-7.09 ~	3.55	0.71	2.84
T-stat		-2.14	1.04	0.00	1.09	0.82	-2.27	2.27	1.00	0.78

Source: Bortolotti and Faccio, 2004.

Table 7. Ultimate Control Rights in Privatized Firms

Data relating to 141 privatized corporations and 141 matching firms are used to construct this table. Control rights is the percentage of voting rights ultimately controlled by the largest controlling shareholder. Government control rights is the percentage of voting rights controlled by the Government as the largest shareholder. Private control rights is the percentage of voting rights controlled by the largest shareholder in firms matching those in which the Government is the largest controlling shareholder. Firms using control enhancing devices denotes the percentage of government-controlled firms (or matching peers) in which the controlling shareholder enhances his/her voting power by using pyramids, multiple control chains and/or dual class share structures. Pyramids occur when the controlling shareholder owns one corporation through another which he/she does not totally own. Firm Y is held through multiple control chains if it has an ultimate owner who controls it via a multitude of control chains, each of which includes at least 5% of the voting rights at each link. Dual class shares occur when firms have outstanding stocks with different voting and/or cash flow rights. a, b, and c denote statistical significance at the .01, .05, and .10 levels, respectively.

		Panel A: Pri	vatized Firms		
Country	Number of firms	Mean	Median	Government Control Rights	Firms using control enhancing devices
End of 1996	141	27.80	19.99	51.27 (N=49)	53.06% (N=49)
End of 2000	141	25.51	16.16	52.18 (N=42)	52.38% (N=42)
Diff '00-'96		-2.29			
T-stat		-1.26			
		Panel B: Ma	Itching Firms		
Country	Number of firms	Mean	Median	Private Control Rights	Firms using control enhancing devices
End of 1996	141	21.10	11.92	15.67 (N=49)	30.61% (N=49)
End of 2000	141	26.37	13.40	17.76 (N=42)	33.33% (N=42)
Diff '00-'96		5.27 ^b			
T-stat		2.13			
	Panel C: Diffe	erence between	Privatized and Ma	tching Firms	

Country	All Firms (Mean)	Government- Controlled Firms
Diff end 1996	6.70 ^b	35.50 ª
T-stat	2.37	9.10
Diff end 2000	-0.86	34.42 ^a
T-stat	-0.28	7.80

Source: Bortolotti and Faccio, 2004.

Table 8. Country Distribution of Privatized Firms by Control Type

Gov't Controlled firms are those whose largest shareholder (at the 10 percent threshold) is a national government (domestic or foreign), local authority (county, municipality, etc.), or government agency. Golden share is a dummy that takes the value of 1 if the Government enjoys special powers or there are statutory constraints in privatized companies.

Country	Obs	Obs. in the Country as % of all Privatizations	Gov't Controlled (as of end '96)	Golden Share (as of end '96)	Gov't Controlled or Golden Share (as of end '96)	Gov't Controlled or Golden Share as % of Privatiz. in the Country (as of end '96)	Gov't Controlled (as of end '00)	Gov't Controlled or Golden Share (as of end '00)	Gov't Controlled or Golden Share as % of Privatiz. in the Country (as of end '00)
		7.00	2	2	2	21.22	0	0	21.22
Austria	11	7.80	9	2	9	81.80	9	9	81.80
Belgium	2	1.40	1	1	2	100.00	1	2	100.00
Denmark	2	1.40	2	1	2	100.00	1	1	50.00
Finland	4	2.80	4	1	4	100.00	4	4	100.00
France	20	14.20	6	5	9	45.00	5	8	40.00
Germany	10	7.10	5	2	5	50.00	5	5	50.00
Greece	2	1.40	2	1	2	100.00	2	2	100.00
Ireland	2	1.40	0	2	2	100.00	0	2	100.00
Italy	12	8.50	6	6	8	66.70	4	7	58.30
Netherlands	3	2.10	1	1	1	33.30	1	1	33.30
Norway	6	4.30	3	2	3	50.00	2	3	50.00
Portugal	9	6.40	1	4	4	44.40	1	4	44.40
Spain	5	3.50	2	2	3	60.00	1	3	60.00
Sweden	3	2.10	2	1	2	66.70	1	1	33.30
Turkey	3	2.10	0	2	2	66.70	0	2	66.70
United Kingdom	24	17.00	0	17	17	70.80	0	17	70.80
Whole sample	118	100	44	50	75	70.9	37	71	64.9

Source: Bortolotti and Faccio, 2004.





Source: Elaborations on Securities Data Corporation.



Figure 2. Privatizations around the World: ranking by revenues (1977-2004)

Source: Elaborations on Securities Data Corporation.



Figure 3. Privatizations around the World: ranking by transactions (1977-2004)

Source: Elaborations on Securities Data Corporation.





Source: Elaborations on Securities Data Corporation.



Figure 5. Privatization in Western Europe: Country Ranking by Revenues (1977-2002)*

Source: Elaboration on Securities Data Corporation, and World Bank.



Figure 6. Privatization in Western Europe: Country Ranking by Revenues relative to GDP *

Source: Elaboration on Securities Data Corporation, and World Bank.



Figure 7. Privatization in Western Europe: Distribution of Revenues by Sector (1977-2004)

Source: Elaborations on Securities Data Corporation



Figure 8. Strategic vs Other Sectors (1977-2004)

Source: Elaboration on Securities Data Corporation.

* "Strategic" Sectors include: Telecommunications, Utilities, Transportation industry, and Petroleum. "Other" Sectors include all the other sectors.



Figure 9. Privatizations in Western Europe: Revenues/GDP and Revenues/SOE (1977-2002)

Source: Elaboration on Securities Data Corporation, and World Bank.

Bibliography

- Alesina, A. and A. Drazen (1991). Why are stabilizations delayed?, *American Economic Review*, 81, 1170-88.
- Baklanoff, E. N., (1986). The state and economy in Portugal: perspectives on corporatism, revolution and incipient privatization, in W.P. Glade, *State Shrinking. A comparative inquiry into Privatization*, Institute of Latin American Studies, Austin, Texas.
- Biais, B., and Perotti, E.C., (2002). Machiavellian privatization, American Economic Review, 92, 240-58.
- Bortolotti, B., and Pinotti, P., (2003). The political economy of privatization, *FEEM Note di Lavoro*, 45.
- Bortolotti, B., De Jong, F., Nicodano, G., and Schindele, I., (2004). Privatization and stock market liquidity, *CEPR Discussion Paper* No. 4449
- Bortolotti, B., D'Souza, J., Fantini, M., Megginson, W.L., (2002). Sources of Performance Improvements in Privatized Firms: A Clinical Study of the Global Telecommunications Industry, *Telecommunications Policy* 26, 243-268.
- Bortolotti, B. and Faccio, M., (2004). Reluctant privatization, *EGCI Working Paper* 40.
- Bortolotti, B., Siniscalco, D., (2004). The Challenges of Privatization: an International Analysis, *Oxford*, Oxford University Press.
- Bortolotti, B., Fantini, M., and Siniscalco, D., (2003). Privatisation around the world: evidence from panel data, *Journal of Public Economics*, 88, 335-366.
- Boutchkova, M.K., and Megginson, W.L., (2000). Privatization and the role of global capital markets, *Financial Management*, 32-76.
- Chiesa, G. and Nicodano, G. (2003), Privatization and Financial Market Development: Theoretical Issues, *FEEM Note di Lavoro*.
- Clarke T., and Pitelis, C., (eds.) (1993). *The political economy of privatization*, London, Routledge.

- Dewenter, K., and Malatesta, P.H., (1997). Public offerings of state-owned and privately owned enterprises: an international comparison, *Journal of Finance*, 57,1659-1679.
- Dumontier, P. and Laurin, C., (2002). The Financial Impacts of the French Government Nationalization - Privatization Strategy, *Working paper*, University of Geneva.
- D'Souza, J. and Megginson, W. L., (1999). The Financial and Operating Performance of Newly Privatized Firms in the 1990s, *Journal of Finance* 54.
- Ellul, A., and M. Pagano (2002). IPO undepricing and after-market liquidity, CSEF University of Salerno, *mimeo*.
- Florio, M., (2004). The Great Divestiture, MIT Press, Cambridge.
- Goldstein, A. (2003). Privatization in Italy 1993 2002: Goals, Institutions, Outcomes, and Outstanding Issues, *CESifo Working Papers, no 912*.
- Gupta, N., (2004). Partial privatization and firm performance, *Journal of Finance*, forthcoming.
- Haggarty, L., and Shirley, M., (1995). Bureaucrats in Business, World Bank Group, Washington.
- Hölmstrom, B. and Tirole, J., (1993). Market Liquidity and Performance Monitoring, *Journal of Political Economy*, 101, 678-709.
- Jenkinson, T., and Ljungqvist, A.P., (2000). *Going public*, London, Oxford University Press.
- Jeronimo, V., Pagán, J. A. and Gökçe S., (2000). Privatization and European Economic and Monetary Union, *Eastern Economic Journal* 26, 321-333.
- Jones, S., Megginson, W.L., Nash, R., and Netter, J., (1999). Share issue privatizations as financial means to political and economic ends, *Journal of Financial Economics*, 53, 217-53.
- Katsoulakos, Y. and Likoyanni, E., (2002). Fiscal and Macroeconomic Effects of Privatization, *Working paper, Athens University of Economics and Business*, Athens.

- Kelohariju, M. Knupfer, G. and Torstila, S., (2004). Do Retail Incentives Work in Privatizations? CEPR Discussion Paper 4612
- La Porta, R., López-de-Silanes, F., Shleifer, A., and Vishny, R.W., (1999). Corporate ownership around the world, *Journal of Finance*, 54, 471-518.
- Lijphart, A., (1999). Patterns of democracy, New Haven and London, Yale University Press.
- Mackenzie, G.A. (1998). The Macroeconomic Impact of Privatization, *IMF Staff Paper*, Vol.45, N° 2. (June) *International Monetary Fund*.

Megginson, W.L., (2003). Macroeconomic effects of privatization, mimeo.

- Megginson, W.L., and Netter, J., (2001). From State to market: a survey of empirical studies on privatiztion, *Journal of Economic Literature*, 39, 321-389.
- Megginson, W.L., Nash, R.C., and van Randenborgh, M., (1994). The financial and operating performance of newly privatised firms: An international empirical analysis, *Journal of Finance*, 49, 403-452.
- Megginson, W.L., Nash, R.C., Netter, J.M., and Poulsen, A.B., (2000). The choice between private and public markets: evidence from privatizations, Working paper, Athens, University of Georgia.
- Newbery, D. and Pollitt, M.G., (1997). "The Restructuring and Privatization of Britain's CEGB Was it Worth it?" *Journal of Industrial Economics* 45, pp. 269-303.
- Nicoletti, G., Scarpetta, S., (2003). Regulation, Productivity and growth: OECD evidence, *ECO/Working Paper N*° 347, OECD.
- Pagano, M., (1993). The floatation of companies and the stock market: a coordination failure model, *European Economic Review*, 37, 1101-1125.
- Price, C. W., and Weyman-Jones, T., (1996). Malmquist Indices of Productivity Change in the UK Gas Industry Before and After Privatization, *Applied Economics* 28, 29-39.
- Sheshinski, E., and Lopez-Calva, L.F., (2000). Privatization and its benefits: theory, evidence, and challenges, in Basu, K., Nayak, P., and Ray, R. (eds.) *Markets and governments*, New Delhi, Oxford University Press.

- Spolaore, E. (2003). Adjustments in Different Government Systems, *Economics and Politics* (forthcoming).
- Subrahmaniam, S. and Titman, S., (1999). The going public decision and the development of financial markets, *Journal of Finance*, 54, 1045-1082.
- Villalonga, B., (2000). Privatization and Efficiency: Differentiating Ownership Effects from Political, Organizational and Dynamic Effects, *Journal of Economic Behavior and Organization* 42, 43-74.
- World Bank, (1995). Bureaucrats in Business: the economics and politics of government ownership, New York, Oxford University Press.

World Bank, (2002). World Development Indicators.

Wright V., (ed.), (1994). Privatisation in Western Europe, London, Pinter Publishers.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/Feem/Pub/Publications/WPapers/default.html

http://www.ssrn.com/link/feem.html

http://www.repec.org

http://agecon.lib.umn.edu

NOTE DI LAVORO PUBLISHED IN 2006

CCMP 2.2006 Fraintian BOSETTI. Carlo CARRARO and Marzio GALEOTTI: Stabilisation Targets, Technical Change and the Macroeconomic Costs of Climate Change Control CCMP 3.2006 Roberto ROSON: Itaracical mage Control Neuroscience STEV 5.2006 Roberto ROSON: Itaracical mage Control Microeconomics versus Ecological Footprint ORIVER COMP 5.2006 Roberto ROSON: Itaracical Research and Readoum Fluctuations in Weather Neurophysics and Readoum Fluctuations in Weather PRCG 7.2006 Michele MORETTO and Pool VLIBORSE: Firm Regulation and Profit Sharing: A Real Option Approach Michele MORETTO and Pool VLIBORSE: Firm Regulation and Profit Sharing: A Real Option Approach Anna AlBERIN and Alme CHLBAH: Discout Rates in Risk v. Money and Money v. Money Tradeoffs CTN 10.2006 Foho CREINA (Svevin): Tourism Specialization and Aussinability: Angreun Policy Analysis Valentina BOSETTI. Mariaester CASSINELL and Alessandro LANZ4 (Issviii): Benchmarking in Tourism Destination, Keeping in Mind the Sustainable Paradigin CCMP 13.2006 Foho SHARTI: Social Optical Applica and the Quality of Economic Development: A Critical Perspective CSRM 12.2000 Foho SHARTIN: Social Optical and factorial Genomic Development: A Critical Perspective CCMP 13.2006 Foho SHARTIN: Social Optical and theconomic Development: A Critic	SIEV	1.2006	Anna ALBERINI: Determinants and Effects on Property Values of Participation in Voluntary Cleanup Programs:
Macroeconomic Costs of Climate Change Control Macroeconomic Costs of Climate Change Control Robbert ROSON: Introducing Imperfect Competition in CGE Models: Technical Aspects and Implications Sergio VERGALLI: The Role of Community in Migration Dynamics SEV 5.2006 Fabio GRAZI. Jeroen CJM, von den BERGH and Piet MEDTFLD: Modeling Spatial Sustainability: Spatial Optimic Disc HENEX and Michael GREENSTONE: The Economic Impacts of Climate Change: Evidence from Arricultural Profits and Random Fluctuations in Weather PRCG 7.2006 Michele MORETTO and Paola 1/LEONESE: Firm Regulation and Profit-Sharing: A Real Option Approach Store 8.2006 Anna ALBERNI and Alme CHILBALL Discount Rates in Risk v. Money and Money v. Money Tradeoffs ON 9.2006 Jon X. EGULH: United We Vale Climate Change: Evidence from German Panel Data Sources NRM 12.2006 Fabio CHINS UNG and Divko DMITRO: A Taxonomy of Myopic Stability Concepts for Hedonic Games NRM 12.2006 Fabio CHINA (txviii): Tourism Specialization and Sustainability: A Long-Run Poley Analysis NRM 12.2006 Fabio CHINA Mark Marine Stability Concepts for Hedonic Games KTHC 12.2006 Fabio Stability Concepts for Hedonic Games KTHC 12.2006 Fabio Stability Concepts for Hedonic Games	CCMP	2.2006	Valentina BOSETTI, Carlo CARRARO and Marzio GALEOTTI: Stabilisation Targets, Technical Change and the
 CCMP 5.2006 Note: Introducing imperfect Competition In COE. Models: Technical Aspects and Impirations KFHC 4.2006 Sergio VERGALL: The Role of Community in Migration Dynamics SIEV 5.2006 VERGALL: The Role of Community in Migrate INETTELD: Modeling Spatial Sustainability: Spatial Welfare Economics versus Ecological Footprint Ollivier DESCHENES and Michael GREENSTONE: The Economic Impacts of Climate Change: Evidence from Agricultural Portis and Random Fluctuations in Weather PRCG 7.2006 Anna AlBERNI and Random Fluctuations in Weather PRCG 7.2006 Anna AlBERNI and Aline CHILBARD Discount Rates in Risk v. Money and Money v. Money Tradeoffs CTN 9.2006 Jone CHIN SUNG and Dinko DMITRO: A Taxonomy of Myonic Subility Concepts for Hedonic Games Fabio CERINA (Iswiii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis Valentina BOSETH, Marinetser CASSINELL and Alessandro LAV24 (Uswiii): Enchmarking in Tourism Destination. Keeping in Mind the Sustainable Paradiguo CCMP 13.2006 Fabio CERINA (Uswiii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis Valentina BOSETH, Marinetser CASSINELL and Alessandro LAV24 (Uswiii): Enchmarking in Tourism Destination. Keeping in Mind the Sustainable Paradiguo CCMP 13.2006 Fabio SABATIN: Stocial Capital, Public Spending and the Quality of Economic Development. The Case of tally KTHC 15.2006 Fabio SABATIN: The Empirics of Social Capital and Economic Development. The Case of tally KTHC 15.2006 Solution MAZANTI: and Roberto ZOBOL: Examining the Factors Influencing Environmental Maratise Control Policies Control Policies Control Policies Control Policies (Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards COMP 20.2006 Sounded LAHRE: A weak Bargaining Set for Contract Choice Polibens Massimiliono MAZANTI and Roberto ZOBOL: Examining the Factors		2 2000	<u>Macroeconomic Costs of Climate Change Control</u>
 KHC 4.2006 Sergio F2RCALLE: <u>The Kole of Community in Migration Dynamics</u> Scope F2RCALLE: <u>Interview of the RERCH and Piet RETVELD</u>: <u>Modeling Spatial Sustainability: Spatial Welfare Economics versus Ecological Footprint</u> CCMP 6.2006 <i>Other DESCHENES and Michael GREENSTONE</i>: <u>The Economic Impacts of Climate Change: Evidence from Agricultural Profits and Random Euctuations in Weather</u> PRCG 7.2006 <i>Michael MORETTO and Pools</i> 241800/XSE: <u>Emonomic Impacts of Climate Change: Evidence from Agricultural Profits and Random Euctuations in Weather</u> PRCG 7.2006 <i>Michael MORETTO and Pools</i> 241800/XSE: <u>Emonomy of Myopic Stability Concepts for Hedonic Games</u> Stav A.2006 <i>Jonx X.EGULI</i>: <u>United We Yote</u> CTN 10.2006 <i>Shao CHIN SUNG and Dinko DIMITRO</i>: <u>A Taxonomy of Myopic Stability Concepts for Hedonic Games</u> NRM 11.2006 <i>Fabio CERNA</i> (hxxviii): <u>Tourism Specialization and Sustainability: A Long Euchmerking in Tourism Destination, Keeping in Mind the Sustainable Paradigm</u> Usen IORBACIT: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources KTHC 14.2006 <i>Fabio SMATINI</i>: <u>Social Capital Public Spending and the Quality of Economic Development: A Critical Perspective Gause of Huly SMATINI: <u>Social Capital and Economic Development</u>: A Critical Perspective Gauserpe DI 1714: <u>Corruption, Evogenous Changes in Incentives and Deterrence</u></u> CCMP 17.2006 <i>Fabio SMATINI</i>: <u>A Weak Barcaining Set for Contract Choice Problems</u> CCMP 20.2006 <i>Massimiliano MAZ2AVII and Roberto ZOBOL</i>: <u>Examining the Faetors Influencing Environmental Informational Greenbouse Gas Emission Reductions in the Presence of Other Environmental Policies</u> Massimiliano MAZZAVII and Roberto ZOBOL: <u>Examining the Faetors Influencing Environmental Information Mazzavi and Ken-ICHI 4KAO: Non-pecuniary Work Incentive and Informatia Environmental Information Mazzavi and Ken-ICHI 4KA</u>	CCMP	3.2006	Roberto ROSON: Introducing Imperfect Competition in CGE Models: Technical Aspects and Implications
SIEV 5.2006 Pable ORAL, Jeroen C.J.M. van den BERGH and Piet RETPELD: Moderation Statial Sustainability: Spalial Mediare Economics versus Ecological Footprint Olivier DESCHENES and Michael GREENSTONE: The Economic Impacts of Climate Change; Evidence from Agricultural Poritis and Random Enclutations in Weather PRCG 7.2006 Michele MORETTO and Paola VALBONES: Firm Regulation and Proft-Sharing: A Real Option Approach SIEV 8.2006 Anna AlbERRIN and Aline CHL4RAIR. Discount Rates in Risk v. Money v. Money Y. Money Tradeoffs CTN 9.2006 Jon X. EGUIN: University Tourism Specialization and Sustainability: A Long-Run Policy Analysis NRM 11.2006 Shao CERRM (txxviii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis NRM 12.2006 Jon X. EGUIN: University Tourism Specialization and Sustainability: A Long-Run Policy Analysis NRM 12.2006 Jons HORBACH: Determinants of Environmental Innovation — New Evidence from German Panel Data Sources CCMP 13.2006 Fabio SABATIN: The Empirics of Social Capital, Aphilis Spending and the Onality of Economic Development: The Case of Italy KTHC 15.2006 Fabio SABATIN: Corruptin, Exoremous Changes in Incentives and Determence CCMP 17.2006 Fabio SABATIN: Corruptin, Exoremous Changes in Incentives and Daterestoce Stave St	KTHC	4.2006	Sergio VERGALLI: The Role of Community in Migration Dynamics
CCMP 6.2006 Olivier DESCHENES and Michael GREENSTONE: The Economic Impacts of Climate Change: Evidence from Agricultural Portis and Random Euclustions in Weather PRCG 7.2006 Michele MORETTO and Paola VALBONESE: Firm Regulation and Profit-Sharine: A real Option Approach SIEV 8.2006 Anna AlbERINI and Aline CHIBAH: Discount Rates in Risk v. Money and Money v. Money Tradeoffs OTN 9.2006 Shao CHIN SUNG and Davino DMITRO: A Taxonomy of Myopic Stability Concepts for Hedonic Games NRM 11.2006 Fabro CERINA (Ixxviii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis NRM 12.2006 Jans HORBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources CCMP 13.2006 Jans HORBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources CINC 14.2006 Fabro SABATIN: The Empirics of Social Capital Apblic Spending and the Ouality of Economic Development: The Case or test CINC 52.2006 Fabro SABATIN: The Empirics of Social Capital Apblic Spending and the Ouality of Economic Development: The Case or test CINC 15.2006 Fabro SABATIN: Social Capital Apblic Spending and the Ouality of Economic Development: The Case or test CINC 15.2006 Fabro SABATINI: The Empirics of Social Capital Apblis Spending and the Ouality and	SIEV	5.2006	Fabio GRAZI, Jeroen C.J.M. van den BERGH and Piet RIETVELD: Modeling Spatial Sustainability: Spatial Welfare Economics versus Ecological Footprint
 PRCG 7.2006 Intervent CHLBARSE: Firm Resultation and Profit-Sharing: A Real Option Approach Anna ALBERINI and Aline CHLBAR: Discount Rates in Risk v. Money and Money v. Money Tradeoffs JON X. EGULI: United We Vote Sharo CHLBARUM and Aline CHLBARD: Discount Rates in Risk v. Money and Money v. Money Tradeoffs JON X. EGULI: United We Vote Sharo CHLBARUM: Social Capital and Economic Development: ChCSMP 13.2006 Jens HORBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources Fabio ShBATINI: Social Capital Deblic Spending and the Quality of Economic Development: ChC Case of Italy KTHC 15.2006 Fabio SABATINI: The Empiriso Social Capital and Economic Development: ChC Case of Italy KTHC 15.2006 Fabio SABATINI: The Empiriso Social Capital and Economic Development: ChC Case of Italy Giuseppe DI VITA: Corruption, Exogenous Changes in Incentives and Determence RCMP 17.2006 Fobio SABATINI: And Marjan W. HOFKES: The Timing of National Greenhouse Gas Emission Reductions in the Presence of Other Environmental Policies Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards SCINT 19.2005 Sondeb L4HIRE: A Weak Barganing Set for Contract Chcice Problems Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations 19.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply Martin GALEGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Devisor FARZIN and Ken-ICHI AKAO: Non-neuniary Work Incentive and Labor Supply Martin GALEGA	CCMP	6.2006	<i>Olivier DESCHENES and Michael GREENSTONE</i> : <u>The Economic Impacts of Climate Change: Evidence from</u> Agricultural Profits and Random Fluctuations in Weather
SIEV 8.2006 Anna ALBERINI and Aline CHLABAI: Discount Rates in Risk v. Money and Money v. Money Tradeoffs CTN 9.2006 Jon X. EGUIA: United We Vote CTN 10.2006 Shao CHIN SUNG and Dinko DMITRO: A Taxonomy of Myopic Stability Concepts for Hedonic Games NRM 11.2006 Fabio CERINA (txviii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis NRM 12.2006 Fabio SABATINI: Mariaster CASSINELLI and Alessandro LANZA (txviii): Benchmarking in Tourism Destination, Keeping in Mind the Sustainable Paradigm Devinting of SabATINI: Social Capital and Economic Development: The Case of Taty KTHC 14.2006 Fabio SABATINI: Tocial Capital and Economic Development: A Critical Perspective CSRM 16.2006 Giuseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence CCMP 17.2006 Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards Somdeb LAHIRI: A Weak Bargaining Set for Contract Choice Problems Somideb LAHIRI: A Weak Bargaining Set for Contract Choice Problems CCMP 20.2006 Marcio GALEOTTI, Markeer AL and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Wartes Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Vash Checentive and Labor Supply CMarcio GALEOTTI, Matteo	PRCG	7.2006	Michele MORETTO and Paola VALBONESE: Firm Regulation and Profit-Sharing: A Real Option Approach
CTN 9.2006 Jon X. EGULI: United We Vote CTN 10.2006 Skao CHIN SUNG and Dinko DMITRO: <u>A Taxonomy of Myopic Stability Concepts for Hedonic Games</u> NRM 11.2006 Fabio CERINA (hxviii): Tourism Specialization and Sustainability: A Long: Run Policy Analysis Valentina BOSETTI, Mariaester CASSINELLI and Alessandro LANZA (hxviii): Benchmarking in Tourism Specialization, Keeping in Mind the Sustainable Paradigm CCMP 13.2006 Jestination, Keeping in Mind the Sustainable Paradigm CCMP 13.2006 Jestination, Keeping in Mind the Sustainable Paradigm CCMP 15.2006 Jest HOBBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources KTHC 15.2006 Fabio SABATIN: Social Capital, Public Specing in Minding Deterence: CCMP 17.2006 Ro b DELLINK and Marian W. HOFKES: The Timing of National Creenhouse Gas Emission Reductions in the Presence of Other Environmental Policies EIM 18.2006 Philippe QUIRNO: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CTN 19.2006 Somdeb LAHIR: A Weak Bargaining Set for Contract Choice Problems Massimiliano MAZZANTI and Roberto 20BOLI: Examining the Factors Influencing Environmental Innovations SIEV 20.2006 Y. Hossein	SIEV	8.2006	Anna ALBERINI and Aline CHIABAI: Discount Rates in Risk v. Money and Money v. Money Tradeoffs
CTN 10.2006 Shao CHIN SUNG and Dinko DIMITRO: A Taxonomy of Myopic Stability: Concepts for Hedonic Games Fabio CERINA (txviii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis NRM 12.2006 Fabio CERINA (txviii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis NRM 12.2006 Valentina BOSETTI, Mariaester CASSINELLI and Alessandro LANZA (txviii): Benchmarking in Tourism Destination, Keeping in Mind the Sustainable Paradigm CCMP 13.2006 Jens HORBACH*, Determinants of Environmental Innovation – New Evidence from German Panel Data Sources KTHC 14.2006 Fabio SABATINI: Social Capital and Economic Development: A Critical Perspective CSRM 16.2006 Giuseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence RCMP 17.2006 Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CTN 19.2006 Somdeb LAHIR: A Weak Bargaining Set for Contract Choice Problems CCMP 21.2006 Philippe QUIRION: Distributional Impacts of Zongrave Under Choice Supply Marxio GALEOTTI, Martiae MANEA and Alessandro LANZA (Sungrave and Labor Supply Marxio GALEOTTI, Martiae MANEA and Alessandro LANZA (Sungrave and Labor Supply Marxio GALEOTTI, Martiae MANEA and Alessandro LANZA (Sungrave and Standards) Stev	CTN	9.2006	Jon X. EGUIA: United We Vote
NRM 11.2006 Fabio CERIVA (txxviii): Courism Specialization and Sustainability: A Long-Run Policy Analysis Valentina BOSETTI, Mariaester CASSINELLI and Alessandro LANZA (txxviii): Benchmarking in Tourism Destination, Keeping in Mind the Sustainable Paradigm CCMP 13.2006 Jens HORBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources KTHC 14.2006 Fabio SABATIN: Social Capital, Public Spending and the Quality of Economic Development: The Case of Italy KTHC 15.2006 Giuseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence RCMP 17.2006 Rob B. DELLINK and Marjan W. HOFKES: The Timing of National Oreenhouse Gas Emission Reductions in the Presence of Other Environmental Policies IEM 18.2006 Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CCMP 20.2006 Incovations Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Imovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply Marsi OGLEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kurnets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment an Natural Resource Extinction	CTN	10.2006	Shao CHIN SUNG and Dinko DIMITRO: A Taxonomy of Myopic Stability Concepts for Hedonic Games
NRM 12.2006 Valentina BOSETTI, Mariaester CASSINELLI and Alessandro LANZA (txxviii): Benchmarking: in Tourism Destination, Keeping in Mind the Sustainable Paradigm CCMP 13.2007 Jers HORBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources KTHC 14.2006 Fabio SABATIN: Social Capital, Public Spending and the Quality of Economic Development: A Critical Perspective CSRM 16.2006 Giuseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence CCMP 17.2006 the Presence of Other Environmental Policies IEM 18.2006 Philippe QURION: Distributional Impacts of Energy-Efficiency Certificates Vs, Taxes and Standards CTN 19.2006 Somdeh LAHIR: A Weak Bargarianing Set for Contract Choice Problems CCMP 20.2006 Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2007 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply Marxio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Autoration Revelopment and Natural Resource in a Finite Time? Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecunia	NRM	11.2006	Fabio CERINA (lxxviii): Tourism Specialization and Sustainability: A Long-Run Policy Analysis
CCMP 13.2006 Jens HORBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources KTHC 14.2006 Fabio SABATIN: Social Capital, Public Spending and the Quality of Economic Development: The Case of Italy KTHC 15.2006 Fabio SABATIN: The Empirics of Social Capital and Economic Development: A Critical Perspective CSRM 16.2006 Giuseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence RCMP 17.2006 the Presence of Other Environmental Policies ERM 18.2006 Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CTN 19.2006 Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CTN 19.2006 Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply Marzio GALEOTTI, Matteo MANERA and Alessandro LANZ4: On the Robustness of Robustness Checks of the Environmental Ruznets Curve RNM 24.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Re	NRM	12.2006	Valentina BOSETTI, Mariaester CASSINELLI and Alessandro LANZA (Ixxviii): <u>Benchmarking in Tourism</u> Destination, Keeping in Mind the Sustainable Paradigm
 KTHC 14.2006 Fabio SABATINI: Social Capital, Public Spending and the Quality of Economic Development: The Case of Italy Fabio SABATINI: The Empirics of Social Capital and Economic Development: A Critical Perspective Ginseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence CCMP 17.2006 Fabio SABATINI: The Empirics of Social Capital and Economic Development: A Critical Perspective Ginseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence Rob B. DELLINK and Marjan W. HOFKES: The Timing of National Greenhouse Gas Emission Reductions in the Presence of Other Environmental Policies EM 18.2006 Philippe QUIRON: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards Somdeb LAHIRI: A Weak Bargaining Set for Contract Choice Problems Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI 4KAO: Non-pecuniary Work Incentive and Labor Supply Marzio GALEOTTI, Mattee MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland KTHC 27.2006 Giovanni BELLA, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Markov-Switching Analysis for the G-7 Countries SIEV 26.2006 Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND:	CCMP	13,2006	Jens HORBACH: Determinants of Environmental Innovation – New Evidence from German Panel Data Sources
 KTHC 15.2006 Fabio SABATINI: The Empirics of Social Capital and Economic Development: A Critical Perspective Giuseppe DI VITA: Corruption, Exogenous Changes in Incentives and Deterrence ROB DELLINK and Margin W. HOPKES: The Timing of National Greenhouse Gas Emission Reductions in the Presence of Other Environmental Policies IEM 18.2006 Fhilippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CCMP 20.2006 Market MIRI: A Weak Bargaining Set for Contract Choice Problems CCMP 20.2006 Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply MCMP 22.2006 Marxing ALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier Approach CCMP 28.2006 Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in A Model with Polluting Emissions Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 Countries S1200 57.2006 Fabio SABATINI: Social Capital and Labour Productivity in Italy CTM 31.2006 Fabio SABATINI: Social Capital and Labour Productivity in Italy CTM 31.2006 Fabio SABATINI: Social Capit	KTHC	14.2006	Fabio SABATINI: Social Capital Public Spending and the Quality of Economic Development: The Case of Italy
CSRM 16.2006 Giuseppe DI VITA: Corruption. Exogenous Changes in Incentives and Deterrence CCMP 17.2006 Rob B. DELLINK and Marjan W. HOFKES: The Timing of National Greenhouse Gas Emission Reductions in the Presence of Other Environmental Policies Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CTN 19.2006 Somdeb LAHIRI: A Weak Bargaining Set for Contract Choice Problems CCMP 20.2006 Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier Approach CCMP 28.2006 Giovanni BELLA: Uniqu	KTHC	15 2006	Fabio SABATINI: The Empirics of Social Capital and Economic Development: A Critical Perspective
 CCMP 17.2006 <i>Rob B. DELLINK and Marjan W. HOFKES</i>: The Timing of National Greenhouse Gas Emission Reductions in the Presence of Other Environmental Policies IEM 18.2006 <i>Philippe QUIRION</i>: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CTN 19.2006 <i>Somdeb LAHIRI</i>: A Weak Bargaining Set for Contract Choice Problems Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 <i>Massimiliano MAZZANTI and Roberto ZOBOLI</i>: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 <i>Massimiliano MAZZANTI and Ken-ICHI AKAO</i>: Non-pecuniary Work Incentive and Labor Supply CCMP 22.2006 <i>Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA</i>: On the Robustness of Robustness Checks of the Environmental Kurnets Curve NRM 23.2006 <i>Y. Hossein FARZIN and Ken-ICHI AKAO</i>: When is it Optimal to Exhaust a Resource in a Finite Time? Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 <i>Lucia VERGANO and Paulo A.L.D. NUNES</i>: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Damy CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Darive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland KTHC 27.2006 <i>Giovanni BELLA</i>: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions Adersandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 Countries Fabio SABATINI: Social Capital and Labour Productivity in Italy ETA 31.2006 <i>Andrea GALLICE</i> (Xxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax Regret Andrea BIGANO and Pau	CSRM	16 2006	Giusenne DI VITA: Corruption Exogenous Changes in Incentives and Deterrence
Item Presence of Other Environmental Policies IEM 18.2006 Philippe QUIRON: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards CCMP 20.2006 Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply MCCMP 22.2006 Marsimiliano MAZZANTI, and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply MCCMP 22.2006 Marsimal Kuznets Curve Name NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? NRM 24.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? NRM 24.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Mon-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland KTHC 27.2006 Giovaani BELLA: Uniqueness and Indeterminacy of Equilibria in a Mo	CCMP	17.2006	Rob B. DELLINK and Marjan W. HOFKES: The Timing of National Greenhouse Gas Emission Reductions in
 Printippe QURION: Distributional Impacts of Energy-Enflicency Certificates Vs. Taxes and Standards Somdeb LAHIR: A Weak Bargaining Set for Contract Choice Problems Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations VI 20206 SiteV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier Approach COMP 28.2006 Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 Countries KTHC 30.2006 Fabio SABATINI: Social Capital and Labour Productivity in Italy ETA 31.2006 Radada BGAO and Paul SHEEHAN: Sassessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to Italy Rinaldo BRAU and Pavis Foruirs Demand to Sardina CTM 34.2006 Fabio SABATINI: Matal Salvatore MONNE Environment, Human Development and Economi	1514	10.0006	the Presence of Other Environmental Policies
CTN 19.2006 Somdeb LAHIRF: <u>A Weak Bargannng Set for Contract Conce Problems</u> CCMP 20.2006 Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply CCMP 22.2006 Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? NRM 24.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland KTHC 27.2006 Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions IEM 29.2006 Markov-Switching Analysis for the G-7 Countries KTHC 30.2006 Fabio SABATINI: Social Capital and Labour Productivity in Italy ETA 31.2006 Andrea GALLACE (Ixxi): Predicting one Shot Play in	IEM	18.2006	Philippe QUIRION: Distributional Impacts of Energy-Efficiency Certificates Vs. Taxes and Standards
CCMP 20.2006 Massimiliano MAZZANIT and Roberto ZOBOLT: Examining the Factors Influencing Environmental Innovations SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply CCMP 22.2006 Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? NRM 24.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland KTHC 27.2006 Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 Countries KTHC 30.2006 Fabio SABATINI: Social Capital and Labour Productivity in Italy ETA 31.2006 Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Medi	CIN	19.2006	Somdeb LAHIRI: A Weak Bargaining Set for Contract Choice Problems
 SIEV 21.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets Curve NRM 23.2006 Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Damy CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland KTHC 27.2006 <i>Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND</i>: Estimating Feedback Effect in Technical Change: A Frontier Approach CCMP 28.2006 <i>Giovanni BELLA:</i> Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 Countries KTHC 30.2006 Fabio SABATINI: Social Capital and Labour Productivity in Italy ETA 31.2006 Andrea GALLICE (Ixxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax Regret Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to Italy NRM 33.2006 <i>Parkash CHANDER and Henry TULKENS</i>: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual Discussion Farkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual Discussion Valeria COSTANTINI and Salvatore MONNF: Environment, Human Development an	CCMP	20.2006	Massimiliano MAZZANTI and Roberto ZOBOLI: Examining the Factors Influencing Environmental Innovations
CCMP22.2006Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the Environmental Kuznets CurveNRM23.2006Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time? Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource ExtinctionSIEV25.2006Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier ApproachCCMP28.2006Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 CountriesKTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyRTH31.2006Andrea GALLICE (txix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax Regret Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Rooute from the Black Sea to ItalyNRM33.2006BRAU and Davide CAO (txixi): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to Sardinia Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual Discussion <tr< td=""><td>SIEV</td><td>21.2006</td><td>Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply</td></tr<>	SIEV	21.2006	Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Work Incentive and Labor Supply
CCMI22.2000Environmental Kuznets CurveNRM23.2006Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time?NRM24.2006Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural ResourceExtinctionExtinctionSIEV25.2006Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An EconomicPerspectiveDanny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments theSIEV26.2006Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental SchemesEvidence from the Rural Environment Protection Scheme in IrelandKTHC27.2006CCMP28.2006Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting EmissionsIEM29.2006Harstov-Switching Analysis for the G-7 CountriesKTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLICE (Ixxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax RegretMRM33.2006Rinaldo BRAU and Davide CAO (Ixxivii): Uncovering the Macrostructure of Tourists' Preferences. A ChoiceExperiment Analysis of Tourism Demand to SardiniaParkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in InternationalEnvironmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Savatore MONNI: Environment, Human Development and Economic GrowthEXA <td< td=""><td>CCMP</td><td>22 2006</td><td>Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the</td></td<>	CCMP	22 2006	Marzio GALEOTTI, Matteo MANERA and Alessandro LANZA: On the Robustness of Robustness Checks of the
 NRM 23.2006 <i>Y. Hossein FARZIN and Ken-ICHI AKAO</i>: When is it Optimal to Exhaust a Resource in a Finite Time? NRM 24.2006 <i>Y. Hossein FARZIN and Ken-ICHI AKAO</i>: Non-pecuniary Value of Employment and Natural Resource Extinction SIEV 25.2006 <i>Lucia VERGANO and Paulo A.L.D. NUNES</i>: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments tr Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland <i>Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND</i>: Estimating Feedback Effect in Technical Change: A Frontier Approach CCMP 28.2006 <i>Giovanni BELLA</i>: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions <i>Alessandro COLOGNI and Matteo MANERA</i>: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 Countries KTHC 30.2006 <i>Fabio SABATINI</i>: Social Capital and Labour Productivity in Italy ETA 31.2006 <i>Andrea GALLICE</i> (Ixxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax Regret <i>Andrea BIGANO and Paul SHEEHAN</i>: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to Italy Rinaldo BRAU and Davide CAO (Ixxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to Sardinia Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environment, Human Development and Economic Growth Ariera RUBINSTEIN (Ixxix): Instinctive and Coonitive Reasonine: A Study of Response Times 	CCIVII	22.2000	Environmental Kuznets Curve
NRM24.2006Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource ExtinctionSIEV25.2006Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Damy CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments trSIEV26.2006Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in IrelandKTHC27.2006Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier ApproachCCMP28.2006Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 CountriesKTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLCE (Ixxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax Regret Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to ItalyNRM33.2006Rinaldo BRAU and Davide CAO (Ixxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionETA35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth ETA <tr< td=""><td>NRM</td><td>23.2006</td><td>Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time?</td></tr<>	NRM	23.2006	Y. Hossein FARZIN and Ken-ICHI AKAO: When is it Optimal to Exhaust a Resource in a Finite Time?
SIEV25.2006Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments trSIEV26.2006Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in IrelandKTHC27.2006Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier ApproachCCMP28.2006Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting EmissionsIEM29.2006Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 CountriesKTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLICE (lixxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax Regret Route from the Black Sea to ItalyNRM33.2006Rinaldo BRAU and Davide CAO (lixxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth Environmental Agreements: A Conceptual Discussion	NRM	24.2006	Y. Hossein FARZIN and Ken-ICHI AKAO: Non-pecuniary Value of Employment and Natural Resource Extinction
Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments trSIEV26.2006Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in IrelandKTHC27.2006Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier ApproachCCMP28.2006Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A 	SIEV	25.2006	Lucia VERGANO and Paulo A.L.D. NUNES: Analysis and Evaluation of Ecosystem Resilience: An Economic Perspective
KTHC27.2006Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier ApproachCCMP28.2006Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting EmissionsIEM29.2006Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 CountriesKTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLICE (Ixxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax RegretIEM32.2006Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to ItalyNRM33.2006Rinaldo BRAU and Davide CAO (Ixxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth Ariel RUBINSTEIN (Ixxix): Instinctive and Cognitive Reasoning: A Study of Response Times	SIEV	26.2006	Danny CAMPBELL, W. George HUTCHINSON and Riccardo SCARPA: Using Discrete Choice Experiments to Derive Individual-Specific WTP Estimates for Landscape Improvements under Agri-Environmental Schemes Evidence from the Rural Environment Protection Scheme in Ireland
CCMP28.2006Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting EmissionsIEM29.2006Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 CountriesKTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLICE (lxxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax RegretIEM32.2006Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to ItalyNRM33.2006Rinaldo BRAU and Davide CAO (lxxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth Ariel RUBINSTEIN (lxxix): Instinctive and Cognitive Reasoning: A Study of Response Times	KTHC	27.2006	Vincent M. OTTO, Timo KUOSMANEN and Ekko C. van IERLAND: Estimating Feedback Effect in Technical Change: A Frontier Approach
IEM29.2006Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov-Switching Analysis for the G-7 CountriesKTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLICE (lxxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax RegretIEM32.2006Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to ItalyNRM33.2006Rinaldo BRAU and Davide CAO (lxxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth ETAETA36.2006Ariel RUBINSTEIN (lxxix): Instinctive and Cognitive Reasoning: A Study of Response Times	CCMP	28.2006	Giovanni BELLA: Uniqueness and Indeterminacy of Equilibria in a Model with Polluting Emissions
KTHC30.2006Fabio SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLICE (lxxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax RegretIEM32.2006Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to ItalyNRM33.2006Rinaldo BRAU and Davide CAO (lxxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth ETAETA36.2006Ariel RUBINSTEIN (lxxix): Instinctive and Cognitive Reasoning: A Study of Response Times	IEM	29.2006	Alessandro COLOGNI and Matteo MANERA: The Asymmetric Effects of Oil Shocks on Output Growth: A Markov Switching Analysis for the G.7 Countries
RTHC50.2006Fable SABATINI: Social Capital and Labour Productivity in ItalyETA31.2006Andrea GALLICE (lxxix): Predicting one Shot Play in 2x2 Games Using Beliefs Based on Minimax RegretIEM32.2006Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to ItalyNRM33.2006Rinaldo BRAU and Davide CAO (lxxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth 	VTUC	20 2006	<u>Markov-switching Analysis for the G-7 Countries</u>
FTAS1.2006Andrea GALLICE (IXXI): <a href="http://www.regitted-international-communication-communicatio-communicatio-communicatio-communi</td> <td></td> <td>30.2000</td> <td><i>Fuolo SADATINI</i>. <u>Social Capital and Labour Productivity in tary</u></td>		30.2000	<i>Fuolo SADATINI</i> . <u>Social Capital and Labour Productivity in tary</u>
IEM 32.2006 Anarea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Splits in the Mediterranean: the Case of the Route from the Black Sea to Italy NRM 33.2006 Rinaldo BRAU and Davide CAO (Ixxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to Sardinia CTN 34.2006 Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual Discussion IEM 35.2006 Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth Ariel RUBINSTEIN (Ixxix): Instinctive and Cognitive Reasoning: A Study of Response Times	EIA	31.2006	Andrea GALLICE (IXXIX): Predicting one Shot Play in 222 Games Using Beners Based on Minimax Regret
NRM33.2006Rinaldo BRAU and Davide CAO (Ixxviii): Uncovering the Macrostructure of Tourists' Preferences. A Choice Experiment Analysis of Tourism Demand to SardiniaCTN34.2006Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual DiscussionIEM35.2006Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth Ariel RUBINSTEIN (Ixxix): Instinctive and Cognitive Reasoning: A Study of Response Times	IEM	32.2006	Andrea BIGANO and Paul SHEEHAN: Assessing the Risk of Oil Spills in the Mediterranean: the Case of the Route from the Black Sea to Italy
CTN 34.2006 Parkash CHANDER and Henry TULKENS: Cooperation, Stability and Self-Enforcement in International Environmental Agreements: A Conceptual Discussion IEM 35.2006 Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth ETA 36.2006 Ariel RUBINSTEIN (lxxix): Instinctive and Cognitive Reasoning: A Study of Response Times	NRM	33.2006	<i>Rinaldo BRAU and Davide CAO</i> (lxxviii): <u>Uncovering the Macrostructure of Tourists' Preferences. A Choice</u> Experiment Analysis of Tourism Demand to Sardinia
CTN 34.2006 I unital Characteristic Competation, Stability, and Sen-Environmental Agreements: A Conceptual Discussion IEM 35.2006 Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth ETA 36.2006 Ariel RUBINSTEIN (lxxix): Instinctive and Cognitive Reasoning: A Study of Response Times			Parkash CHANDER and Henry TULKENS: Cooperation Stability and Self-Enforcement in International
IEM 35.2006 Valeria COSTANTINI and Salvatore MONNI: Environment, Human Development and Economic Growth ETA 36.2006 Ariel RUBINSTEIN (lxxix): Instinctive and Cognitive Reasoning: A Study of Response Times	CTN	34.2006	Environmental Agreements: A Concentual Discussion
ETA 36.2006 Ariel RUBINSTEIN (Ixxix): Instinctive and Cognitive Reasoning: A Study of Response Times	IEM	35 2006	Valeria COSTANTINI and Salvatore MONNI: Environment Human Development and Economic Growth
	ETA	36.2006	Ariel RUBINSTEIN (lxxix): Instinctive and Cognitive Reasoning: A Study of Response Times

ETA	37.2006	Maria SALGADeO (lxxix): Choosing to Have Less Choice
ETA	38,2006	Justina A.V. FISCHER and Benno TORGLER: Does Envy Destroy Social Fundamentals? The Impact of Relative
2111	50.2000	Income Position on Social Capital
ETA	39.2006	Empirical Panel Analysis
CCMP	40.2006	Alberto GAGO, Xavier LABANDEIRA, Fidel PICOS And Miguel RODRÍGUEZ: <u>Taxing Tourism In Spain</u> : Results and Recommendations
11714	41.0000	Karl van BIERVLIET, Dirk Le ROY and Paulo A.L.D. NUNES: An Accidental Oil Spill Along the Belgian
IEW	41.2000	Coast: Results from a CV Study
CCMP	42.2006	Rolf GOLOMBEK and Michael HOEL: Endogenous Technology and Tradable Emission Quotas
KTHC	43.2006	Strategy and Organization
CCMP	44.2006	Alvaro CALZADILLA, Francesco PAULI and Roberto ROSON: <u>Climate Change and Extreme Events: An</u> Assessment of Economic Implications
SIEV	45.2006	M.E. KRAGT, P.C. ROEBELING and A. RUIJS: Effects of Great Barrier Reef Degradation on Recreational
NRM	46.2006	C. GIUPPONI, R. CAMERA, A. FASSIO, A. LASUT, J. MYSIAK and A. SGOBBI: <u>Network Analysis, Creative</u>
KTUC	47 2006	System Modelling and DecisionSupport: <i>The NetSyMoD Approach</i> Walter F. LALICH (lxxx): <u>Measurement and Spatial Effects of the Immigrant Created Cultural Diversity in</u>
KIHC	47.2006	Sydney
KTHC	48.2006	Elena PASPALANOVA (lxxx): Cultural Diversity Determining the Memory of a Controversial Social Event
KTHC	49.2006	Ugo GASPARINO, Barbara DEL CORPO and Dino PINELLI (Ixxx): <u>Perceived Diversity of Complex</u> Environmental Systems: Multidimensional Measurement and Synthetic Indicators
KTHC	50.2006	Aleksandra HAUKE (lxxx): Impact of Cultural Differences on Knowledge Transfer in British, Hungarian and Polish Enterprises
KTHC	51.2006	Katherine MARQUAND FORSYTH and Vanja M. K. STENIUS (lxxx): <u>The Challenges of Data Comparison and</u> Varied European Concepts of Diversity
KTHC	52.2006	<i>Gianmarco I.P. OTTAVIANO and Giovanni PERI</i> (lxxx): <u>Rethinking the Gains from Immigration: Theory and</u> Evidence from the U.S.
KTHC	53.2006	Monica BARNI (lxxx): From Statistical to Geolinguistic Data: Mapping and Measuring Linguistic Diversity
KTHC	54.2006	Lucia TAJOLI and Lucia DE BENEDICTIS (lxxx): Economic Integration and Similarity in Trade Structures
KTHC	55.2006	Suzanna CHAN (Ixxx): "God's Little Acre" and "Belfast Chinatown": Diversity and Ethnic Place Identity in Belfast
KTHC	56.2006	Diana PETKOVA (lxxx): Cultural Diversity in People's Attitudes and Perceptions
KTHC	57.2006	John J. <i>BETANCUR</i> (lxxx): From Outsiders to On-Paper Equals to Cultural Curiosities? The Trajectory of Diversity in the USA
KTHC	58.2006	Kiflemariam HAMDE (lxxx): Cultural Diversity A Glimpse Over the Current Debate in Sweden
KTHC	59.2006	Emilio GREGORI (lxxx): Indicators of Migrants' Socio-Professional Integration
KTHC	60.2006	Christa-Maria LERM HAYES (lxxx): Unity in Diversity Through Art? Joseph Beuys' Models of Cultural Dialogue
KTHC	61.2006	Sara VERTOMMEN and Albert MARTENS (lxxx): Ethnic Minorities Rewarded: Ethnostratification on the Wage Market in Belgium
KTHC	62.2006	Nicola GENOVESE and Maria Grazia LA SPADA (lxxx): Diversity and Pluralism: An Economist's View
KTHC	63.2006	Carla BAGNA (lxxx): Italian Schools and New Linguistic Minorities: Nationality Vs. Plurilingualism. Which
ктис	64 2006	Ways and Methodologies for Mapping these Contexts? Vadran OMANOVIĆ (Ixxx): Understanding "Diversity in Organizations" Paradigmatically and Methodologically
	0 4 .2000	Mila PASPALANOVA (lxxx): Identifying and Assessing the Development of Populations of Undocumented
KTHC	65.2006	Migrants: The Case of Undocumented Poles and Bulgarians in Brussels
KTHC	66.2006	Roberto ALZETTA (lxxx): Diversities in Diversity: Exploring Moroccan Migrants' Livelihood in Genoa
KTHC	67.2006	Monika SEDENKOVA and Jiri HORAK (lxxx): <u>Multivariate and Multicriteria Evaluation of Labour Market</u> <u>Situation</u>
KTHC	68.2006	<i>Dirk JACOBS and Andrea REA</i> (lxxx): <u>Construction and Import of Ethnic Categorisations: "Allochthones" in</u> The Netherlands and Belgium
KTHC	69.2006	Eric M. USLANER (lxxx): Does Diversity Drive Down Trust?
KTHC	70.2006	Paula MOTA SANTOS and João BORGES DE SOUSA (lxxx): <u>Visibility & Invisibility of Communities in Urban</u> <u>Systems</u>
ETA	71.2006	Rinaldo BRAU and Matteo LIPPI BRUNI: Eliciting the Demand for Long Term Care Coverage: A Discrete Choice Modelling Analysis
CTN	72.2006	Dinko DIMITROV and Claus-JOCHEN HAAKE: Coalition Formation in Simple Games: The Semistrict Core
CTN	73.2006	Ottorino CHILLEM, Benedetto GUI and Lorenzo ROCCO: On The Economic Value of Repeated Interactions Under Adverse Selection
CTN	74.2006	Sylvain BEAL and Nicolas QUÉROU: Bounded Rationality and Repeated Network Formation
CTN	75.2006	Sophie BADE, Guillaume HAERINGER and Ludovic RENOU: Bilateral Commitment
CTN	76.2006	Andranik TANGIAN: Evaluation of Parties and Coalitions After Parliamentary Elections
CTN	77.2006	<i>Rudolf BERGHAMMER, Agnieszka RUSINOWSKA and Harrie de SWART</i> : <u>Applications of Relations and</u> <u>Graphs to Coalition Formation</u>
CTN	78.2006	Paolo PIN: Eight Degrees of Separation
CTN	79.2006	Roland AMANN and Thomas GALL: How (not) to Choose Peers in Studying Groups

CTN	80.2006	Maria MONTERO: Inequity Aversion May Increase Inequity
CCMP	81.2006	Vincent M. OTTO, Andreas LÖSCHEL and John REILLY: Directed Technical Change and Climate Policy
CSPM	82 2006	Nicoletta FERRO: Riding the Waves of Reforms in Corporate Law, an Overview of Recent Improvements in
CSIM	82.2000	Italian Corporate Codes of Conduct
CTN	83 2006	Siddhartha BANDYOPADHYAY and Mandar OAK: Coalition Governments in a Model of Parliamentary
env	03.2000	Democracy
PRCG	84,2006	Raphaël SOUBEYRAN: Valence Advantages and Public Goods Consumption: Does a Disadvantaged Candidate
inco	01.2000	Choose an Extremist Position?
CCMP	85.2006	Eduardo L. GIMENEZ and Miguel RODRIGUEZ: Pigou's Dividend versus Ramsey's Dividend in the Double
		Dividend Literature
CCMP	86.2006	Andrea BIGANO, Jacqueline M. HAMILTON and Richard S.J. TOL: The Impact of Climate Change on
KTUO	97 2006	Domestic and International Tourism: A Simulation Study
KIHC	87.2006	Fablo SABATINI: Educational Qualification, work Status and Entrepreneurship in Italy an Exploratory Analysis
CCMP	88.2006	<i>Richard S.J. TOL:</i> The Polluter Pays Principle and Cost-Benefit Analysis of Climate Change: An Application of <i>Fund</i>
		<u>Funa</u> Philippe TILLKENS and Henry TILLKENS: The White House and The Kyoto Protocol: Double Standards on
CCMP	89.2006	Uncertainties and Their Consequences
		Andrea M LEITER and Gerald L PRUCKNER: Proportionality of Willingness to Pay to Small Risk Changes –
SIEV	90.2006	The Impact of Attitudinal Factors in Scope Tests
PRCG	91 2006	Ranhäel SOUBEYRAN: When Inertia Generates Political Cycles
CCMP	92 2006	Alireza NAGHAVI: Can R&D-Inducing Green Tariffs Replace International Environmental Regulations?
CCMI	72.2000	Yaviar PALITREL: Reconsidering The Impact of Environment on Long-Pun Growth When Pollution Influences
CCMP	93.2006	Health and Agents Have Finite-I ifetime
		Corrado Di MARIA and Edwin van der WERF: Carbon Leakage Revisited: Unilateral Climate Policy with
CCMP	94.2006	Directed Technical Change
	05 000 6	Paulo A.L.D. NUNES and Chiara M. TRAVISI: Comparing Tax and Tax Reallocations Payments in Financing
ССМР	95.2006	Rail Noise Abatement Programs: Results from a CE valuation study in Italy
CCMD	06 2006	Timo KUOSMANEN and Mika KORTELAINEN: Valuing Environmental Factors in Cost-Benefit Analysis Using
CCMP	96.2006	Data Envelopment Analysis
KTHC	07 2006	Dermot LEAHY and Alireza NAGHAVI: Intellectual Property Rights and Entry into a Foreign Market: FDI vs.
KIIIC	97.2000	Joint Ventures
CCMP	98 2006	Inmaculada MARTÍNEZ-ZARZOSO, Aurelia BENGOCHEA-MORANCHO and Rafael MORALES LAGE: The
cenn	90.2000	Impact of Population on CO2 Emissions: Evidence from European Countries
PRCG	99.2006	Alberto CAVALIERE and Simona SCABROSETTI: Privatization and Efficiency: From Principals and Agents to
		Political Economy
NRM	100.2006	Khaled ABU-ZEID and Sameh AFIFI: Multi-Sectoral Uses of Water & Approaches to DSS in Water
		Management in the NOSTRUM Partner Countries of the Mediterranean
NRM	101.2006	for Water Pascuras Management in the Mediterranean Pasin
		Kerstin RONNERERGER Maria REPRITTELLA Francesco ROSELLO and Richard S I TOL: Klum@Gtan:
CCMP	102.2006	Introducing Biophysical Aspects of Land-Use Decisions Into a General Equilibrium Model A Coupling
		Experiment
VELIC	102 2006	Avner BEN-NER, Brian P. McCALL, Massoud STEPHANE, and Hua WANG: Identity and Self-Other
KTHC	103.2006	Differentiation in Work and Giving Behaviors: Experimental Evidence
SIEV	104 2006	Aline CHIABAI and Paulo A.L.D. NUNES: Economic Valuation of Oceanographic Forecasting Services: A Cost-
SIEV	104.2000	Benefit Exercise
NRM	105 2006	Paola MINOIA and Anna BRUSAROSCO: Water Infrastructures Facing Sustainable Development Challenges:
	105.2000	Integrated Evaluation of Impacts of Dams on Regional Development in Morocco
PRCG	106.2006	Carmine GUERRIERO: Endogenous Price Mechanisms, Capture and Accountability Rules: Theory and
		Evidence
CCMP	107.2006	Richard S.J. TOL, Stephen W. PACALA and Robert SOCOLOW: <u>Understanding Long-Term Energy Use and</u>
		<u>Carbon Dioxide Emissions in the Usa</u>
NRM	108.2006	Carles MANERA and Jaume GARAU TABERNER: The Recent Evolution and Impact of Tourism in the
DDCC	100 2006	Mediterranean: The Case of Island Regions, 1990-2002
PRCG	109.2006	Lake EQOT (here): Marging Diversity in Miles, Historical Approaches to Lake Lucker Lucker
KTHC	110.2006	John FOOT (IXXX): Mapping Diversity in Milan. Historical Approaches to Urban Immigration
KTHC	111.2006	Donatella CALABI: Foreigners and the City: An Historiographical Exploration for the Early Modern Period
IEM	112.2006	Andrea BIGANO, Francesco BOSELLO and Giuseppe MARANO: Energy Demand and Temperature: A
		Dynamic Panel Analysis
SIEV	113.2006	Anna ALBERINI, Stefania IONIN, Margherita TURVANI and Aline CHIABAI: Paying for Permanence: <u>Public</u>
		<u>Presences for Contaminated Site Cleanup</u>
CCMP	114.2006	riverkananaa MUKHERJEE and Dirk I.G. KUBBELKE: Global Climate Change, Technology Transfer and Trade with Complete Specialization
NRM	115 2004	<u>Trade with Complete Specialization</u> Clive LIPCHIN: A Future for the Dead Sea Basin: Water Culture among Israelis, Palastinians and Iordanians
1 11/11/1	113.2000	Rarbara RUCHNER Carlo CARRARO and A Danny FILERMAN. The Allocation of European Union
CCMP	116.2006	Allowances: Lessons Unifying Themes and General Principles
ССМР	117 2006	Richard S.J. TOL: Carbon Dioxide Emission Scenarios for the Usa
	117.2000	

NRM	118.2006	Isabel CORTÉS-JIMÉNEZ and Manuela PULINA: A further step into the ELGH and TLGH for Spain and Italy
SIEV	119.2006	<i>Beat HINTERMANN, Anna ALBERINI and Anil MARKANDYA</i> : <u>Estimating the Value of Safety with Labor</u> Market Data: Are the Results Trustworthy?
SIEV	120.2006	Elena STRUKOVA, Alexander GOLUB and Anil MARKANDYA: <u>Air Pollution Costs in Ukraine</u>
CCMP	121.2006	Massimiliano MAZZANTI, Antonio MUSOLESI and Roberto ZOBOLI: A Bayesian Approach to the Estimation
		of Environmental Kuznets Curves for CO ₂ Emissions
ETA	122.2006	Jean-Marie GRETHER, Nicole A. MATHYS, and Jaime DE MELO: Unraveling the World-Wide Pollution
		Haven Effect
KTHC	123.2006	Sergio VERGALLI: Entry and Exit Strategies in Migration Dynamics
PRCG	124.2006	Bernardo BORTOLOTTI and Valentina MILELLA: Privatization in Western Europe Stylized Facts, Outcomes
		and Open Issues

(lxxviii) This paper was presented at the Second International Conference on "Tourism and Sustainable Economic Development - Macro and Micro Economic Issues" jointly organised by CRENoS (Università di Cagliari and Sassari, Italy) and Fondazione Eni Enrico Mattei, Italy, and supported by the World Bank, Chia, Italy, 16-17 September 2005.

(lxxix) This paper was presented at the International Workshop on "Economic Theory and Experimental Economics" jointly organised by SET (Center for advanced Studies in Economic Theory, University of Milano-Bicocca) and Fondazione Eni Enrico Mattei, Italy, Milan, 20-23 November 2005. The Workshop was co-sponsored by CISEPS (Center for Interdisciplinary Studies in Economics and Social Sciences, University of Milan-Bicocca).

(lxxx) This paper was presented at the First EURODIV Conference "Understanding diversity: Mapping and measuring", held in Milan on 26-27 January 2006 and supported by the Marie Curie Series of Conferences "Cultural Diversity in Europe: a Series of Conferences.

(lxxxi) This paper was presented at the EAERE-FEEM-VIU Summer School on "Computable General Equilibrium Modeling in Environmental and Resource Economics", held in Venice from June 25th to July 1st, 2006 and supported by the Marie Curie Series of Conferences "European Summer School in Resource and Environmental Economics".

	2006 SERIES
ССМР	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
SIEV	Sustainability Indicators and Environmental Valuation (Editor: Anna Alberini)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
KTHC	Knowledge, Technology, Human Capital (Editor: Gianmarco Ottaviano)
IEM	International Energy Markets (Editor: Matteo Manera)
CSRM	Corporate Social Responsibility and Sustainable Management (Editor: Giulio Sapelli)
PRCG	Privatisation Regulation Corporate Governance (Editor: Bernardo Bortolotti)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)
CTN	Coalition Theory Network